1993 SNA Rev 1 – Preliminary draft

List of chapters

Annex II

Accounts

Part 1:	Introduction and overview
1	Introduction
2	Overview
Part 2:	Accounting conventions
3	Flows, stocks and accounting rules
4	Institutional units and sectors
5	Establishments and industries
Part 3:	The sequence of accounts
6	The production account
7	The distribution of income accounts
8	The redistribution of income accounts
9	The use of income accounts
10	The capital account
11	The financial account
12	Other changes in assets accounts
13	The balance sheet
Part 4: Inte	egration and elaboration of the accounts
14	Typical summary presentations
15	Supply and use tables and input-output
16	Price and volume measures
17	Cross-cutting and other special issues
18	Population and labour inputs
19	The role of capital services in the accounts
Part 5: Inte	erpretation of the accounts and extensions
20	Measuring corporate activity
21	The government and public sectors
22	Non-profit institutions in the System
23	Households
24	The informal sector
25	The rest of the world account (external transactions account)
26	The link to financial and monetary statistics
27	Satellite accounts and other extensions of the System
Annex I:	Classifications

General Notes to the Readers of the Preliminary Draft Chapters

The preliminary draft of chapters of the 1993 Rev1 are posted on the UN web site at http://unstats.un.org/unsd/sna1993/draftingPhase/ChapterIssueMatrix.asp for a period of 60 days during which comments from anyone interested in the SNA are welcome. It is hoped that members of the Advisory Expert Group (AEG) and national accounts staff in countries will be particularly willing and eager to offer comments. The preliminary draft will be revised in light of these comments to form the final draft. Because of the tight time table to produce the final draft, readers are informed that deadlines for comments cannot be extended. The deadline for each chapter is given on the page specific to each chapter

A comparison between each draft chapter and the corresponding chapter (if it existed) of the 1993 SNA is available. See the second section of this web page entitled "Reference text of the 1993 SNA"

In the draft chapters, definitions of national accounting concepts and terms are in bold italics. These definitions will also be extracted to form a glossary of key terms.

References to other chapters and other paragraphs in the updated 1993 SNA are highlighted in yellow but have not yet been updated. Final cross references will be inserted when the paragraph numbers have been finalized.. References to other manuals are highlighted in turquoise If there are other citations that would be useful, please submit these suggestions along with comments..

The tables in the text have been reformatted. They appear inset into portrait pages. Most tables appear on successive left- and right-hand side pages. Within each chapter, a table may appear in several places with alternative detail in each appearance to draw attention to the section currently being discussed in the text. The full detail of each table will appear in an annex and this annex will be available electronically in Excel format so that users can see the embedded formula used to derive the tables.

The code numbers for transactions, flows, sectors, etc. have been suppressed in both the text and the tables for the moment. Discussions continue about changes to the existing codes for the 1993 SNA and until this is settled it is difficult to envisage the format of codes for a somewhat amended hierarchy. They will be inserted at a later stage.

The data values in the tables are being reviewed. Where new entries have been added, at present the entries are usually blank but this needs to be rectified. In general, (excepting this case) a blank entry signifies that no entry is possible. A possible entry appears as a zero. The occurrence of blanks and zeroes is being verified also.

Chapter 17: Note by the editor

This chapter covers much of the material new in Rev 1 and aims to bring together all considerations of each of the items so that the reader does not have to skip from chapter to chapter to piece together the complete treatment of any of the issues.

The chapter has had a long gestation and different people have looked at some parts and offered substantive comments. In particular the Eurostat/ECB group working on the guidelines for a supplementary table on pensions have commented on part 3 dealing with pensions. Reflecting the fact that parts of the chapter have appeared in different forms to different people, the current version is labeled V2-2 to distinguish it from earlier versions.

Part 1 concerns insurance including social insurance other than pensions. It is based loosely on annex IV of the 1993 text but includes, for example, new advice on the treatment of reinsurance and a better description of social security when government uses social security as a vehicle for assuming the responsibilities of other employers to their employees. Part 1 deals only with non-pension social benefits; pensions are covered in part 3

Part 2 describes the agreed treatment for standardised guarantees which resembles the treatment of insurance in several aspects.

Part 3 deals with pensions. It explains at length and with detailed numerical examples the detailed recording of pension entitlements under defined contribution and defined benefit pension plans. It also includes the proposed supplementary table that gives flexibility to countries not to include all liabilities for pensions in the main accounts while still providing the information for analysts.

Part 4 describes flows associated with financial assets and liabilities. It subsumes the material in annex III of the 1993 text but is more extensive and gives a detailed description of the services and other flows associated with transactions in all forms of financial assets and liabilities.

Part 5 brings together all discussion on contracts, leases and licences that emerged from the Canberra II Group and have been further discussed by the AEG and ISWGNA.

Anne Harrison

27 June 2007

	t 1 The treatment of insurance and social insurance sions	_
۸.	Coverage of this part	
	1. Direct insurance	6
	2. Reinsurance	
	3. Social insurance schemes	7
	Social benefits	
	Individual policies qualifying as social insurance	9
	4. The units involved	9
	Multi-employer schemes	
	Coverage of social security	
	The measurement of output of direct insurance	
	1. Property income attributed to policyholders/beneficiaries	
	2. Non-life insurance claims	
	3. Defining insurance output	11
	4. Net premiums/contributions and consumption of insurance services	
	5. Recording claims	
	6. Insurance services provided to and from the rest of the world	
	The measurement of reinsurance	
	The recording of non-life insurance and social insurance other than pensions	
	1. Social security schemes of government (excluding pensions)	
	2. Unfunded social insurance schemes operated by employers	15
	3. Individual non-life insurance other than policies qualifying as social insurance	16
	4. Individual life insurance other than policies qualifying as social insurance and an	nuities
	5. Funded social insurance other than pensions	
	6. A note on the tables	19
	Annuities	23
ar	5. Funded social insurance other than pensions.6. A note on the tables.	181923
	Standardised loan guarantee schemes	
	2. Loan guarantees provided by government	
Par 3.	t 3 The treatment of pensions in the System Types of pension schemes	28
	Social security pensions	
	• •	
	Employment related pension schemes other than social security	
	1. Defined contribution pension schemes	30
·		30

	Transactions recorded for a defined benefit pension scheme	35	
	Defined benefit pension schemes operated by other than employers		
	The relationship between the employer and the pension fund	36	
	3. A numerical example	36	
	4. Other changes in the value of pension entitlements	40	
	Defined contribution pension scheme	40	
	Defined benefit pension scheme		
	The issue of promotions	40	
	5. Transferring pension entitlements		
	6. A note on the tables	42	
J.	The special case of government providing pensions via social security	43	
	rt 4 The recording of flows associated with financial bilities 46	assets	and
K.	Introduction	46	
	1. Charging for financial services		
	2. Explicit and implicit charges		
	3. Investment income associated with financial instruments		
	4. Holding gains and losses on financial instruments		
L.	Recording flows in financial instruments		
	1. Monetary gold		
	2. SDRs.		
	3. Currency		
	4. Deposits and loans		
	5. Debt securities		
	Service charges associated with securities		
	Valuing securities		
	Determining interest flows on bills and bonds		
	Interest on bills and similar instruments		
	Interest on bonds and debentures		
	Zero-coupon bonds	51	
	Other bonds, including deep-discounted bonds		
	Index linked securities		
	6. Equity and investment fund shares		
	7. Financial derivatives		
	8. Employee stock options	53	
	9. Other accounts receivable/payable		
Pai	rt 5 Contracts, leases and licences	56	
M.	Introduction		
N.	Leases		
	1. Operating leases		
	2. Financial leases		
0	3. Resource leases		
O.	Licences and permits to use a natural resource	58	
٠.	1 771 (4 1 1 1 2)	70	
··	 The "mobile phone" treatment of licences or permissions to use a natural resourt Radio spectra 		

	3. Land	61
	4. Timber	61
	5. Fish	61
	6. Water	62
	7. Mineral deposits	62
P.	Sharing assets	
Q.	Permissions to undertake a specific activity	
	1. Permits issued by government	
	2. Permits issued by other units	
R.	Contracts for future production	64
S.	Leases as assets	65
T.	Other considerations	
	1. Time share arrangements	
	2. Lost deposits	66
	2. 200 . 44 p 002.0	

3

Chapter 17: Cross-cutting and other special issues

Part 1 The treatment of insurance and social insurance other than pensions

A. Coverage of this part

- 17.1 At its simplest, an insurance policy is an agreement between an insurance corporation and another institutional unit, called the policy holder. Under the agreement, the policy holder makes a payment (a premium) to the insurance corporation and, if or when a specified event occurs, the insurance corporation makes a payment (claim) to the policy holder. In this way, the policy holder protects itself against certain forms of risk; by pooling the risks the insurance corporation aims to receive more from the receipt of premiums than it has to pay out as claims. However, simply recording the actual premiums and claims paid in the accounts of the System would not reflect the links between premiums and claims. Instead, some actual transactions are decomposed and others are imputed in order to bring out the underlying economic processes actually taking place.
- 17.2 The most common form of insurance is called direct insurance whereby the policy is issued by an insurance corporation to another type of institutional unit but an important form of insurance is provided by one insurance corporation to another insurance corporation. This sort of insurance is called reinsurance.
- 17.3 In addition to insurance policies issued on a case-by-case basis by insurance corporations, an important set of contingencies are covered on a collective basis by social insurance schemes. The most important benefit covered by social insurance schemes is income in retirement (pensions) but many other contingencies are covered also.

- 17.4 This part of chapter 17 is concerned with direct insurance, reinsurance and social insurance schemes other than provision of pensions. It attempts to bring together all the entries in the accounts connected with insurance and explain their interconnection. A separate part deals with pensions.
- 17.5 Defining some of the terms peculiar to the insurance industry is a helpful preliminary to further discussion. For direct insurance, the term premiums is used for payment to the insurance corporation; payments by the insurance corporation are called either claims or benefits. For social insurance, payments to the social insurance scheme are referred to as contributions and payments by the scheme are referred to as benefits. The actual premium is the amount payable to the direct insurer or reinsurer to secure insurance cover for a specific event over a stated time period. Cover is frequently provided for one year at a time with the premium due to be paid at the outset though cover may be provided for shorter (or longer) periods and the premium may be payable in instalments, for example monthly. Contributions to social insurance schemes are frequently paid on a monthly or even more frequent basis as they are often made directly when wages and salaries are payable.
- 17.6 The premium earned is the part of the actual premium that relates to cover provided in the accounting period. For example, if an annual policy with a premium of 120 units comes into force on April 1 and accounts are being prepared for a calendar year, the premium earned in the calendar year is 90. The

unearned premium is the amount of the actual premium received that relates to the period past the accounting point. In the example just given, at the end of the accounting period there will be an unearned premium of 30, intended to provide cover for the first three months of the next year. A claim (benefit) is the amount payable to the policy holder by the direct insurer or reinsurer in respect of an event covered by the policy occurring in the period for which the policy is valid. Claims become due when the event occurs, even if the payment is made some time later. Claims that become due are described as claims incurred. In some contested cases the delay between the occurrence of the event giving rise to the claim and the settlement of the claim may be several years. Claims outstanding cover claims that have not been reported, have been reported but are not yet settled or have been both reported and settled but not yet paid.

1. Direct insurance

There are two types of direct insurance, life and non-life insurance. Life insurance is an activity whereby a policyholder makes regular payments to an insurer in return for which the insurer guarantees to provide the policyholder with an agreed sum, or an annuity, at a given date or earlier if the policyholder dies beforehand. The sum payable under the policy (benefit) may be fixed or may vary to reflect the income earned from the investment of premiums during the period for which the policy operates. For policies with varying returns, the terms "with-profits" life insurance or endowment policy are generally used. Although the date and sum may be variable, a claim is always paid in respect of a life policy. Non-life insurance covers all other risks, accidents, sickness, fire, etc. A policy that provides a benefit in the case of death within a given period but in no other circumstances, usually called term insurance, is regarded as non-life insurance because, as with other nonlife insurance, a claim is payable only if a specified contingency occurs and not otherwise. In practice, because of the way in which insurance corporations keep their accounts, it may not always be possible to separate term insurance from other life insurance. In these circumstances, term insurance may have to be treated in the same way as life insurance for purely practical reasons.

- 178 What life and non-life insurance have in common is that they both involve spreading risk. Insurers receive many (relatively) small premiums payments of regular policyholders and pay much larger sums to claimants when the contingencies covered by the policy occur. For non-life insurance, the risks are spread over the whole population that takes out the insurance policies. For example, insurance corporation determines the premiums charged for vehicle insurance in a year by relating them to the amount of claims it expects to pay on vehicle insurance in the same Typically, the number of claimants is much smaller than the number of policyholders. For an individual policyholder there is no relationship between the premiums paid and the claims received, even in the long run, but the insurance corporation establishes such a relationship for every class of non-life insurance on a yearly basis. For life insurance, a relationship between premiums and claims over time is important both to the policyholders and to the insurance corporation. For someone taking out a life policy, the benefits to be received are expected to be at least as great as the premiums paid up until the benefit (claim) is due and can be seen as a form of saving. The insurance corporation must combine this aspect of a single policy with the actuarial calculations about the insured population concerning life expectancy (including the risks of fatal accidents) when determining the relationship between the levels of premiums and benefits. Further, in the interval between the receipt of premiums and the payment of claims, the insurance corporation earns income from investing the premiums received. This income also affects the levels of premiums and benefits set by the insurance corporations.
- 17.9 Despite the similarity of the activity of life and non-life insurance, there are significant differences between them that lead to different types of entries in the accounts of the System.
- 17.10 One way in which a regular income stream can be obtained in return for an up-font payment of a lump sum is via an annuity. Annuities are usually offered by life insurance corporations and so a discussion of the recording for annuities in the System is given at the end of this part.

2. Reinsurance

- 17.11 Just as an individual institutional unit protects itself against the financial consequences of loss or damage, so an insurance corporation may also protect itself against an unexpectedly large number of claims, or exceptionally heavy claims, by taking out a reinsurance policy with another insurance corporation. All insurance corporations take out some form of reinsurance but there tend to be a few large corporations that specialise in issuing reinsurance policies. Because these corporations are concentrated in a few financial centres, many of the flows associated with reinsurance involve transactions with the rest of the world. It is common for reinsurers to take out reinsurance policies with other insurance corporations to spread their risks further. This sort of reinsurance is called retrocession.
- 17.12 Reinsurance policies are most common for nonlife policies but may also apply to life insurance policies. There are of two types of reinsurance, proportionate reinsurance and excess of loss reinsurance. Under a proportionate reinsurance contract, the reinsurer accepts an agreed proportion of the risks; this proportion of the premiums is "ceded" to the reinsurer who then meets the same proportion of the claims. In this case, any reinsurance commission paid by the reinsurer to the policy holder (either a direct insurer or another reinsurer) is treated as a reduction in reinsurance premiums payable. In excess of loss reinsurance, the reinsurer undertakes to pay all losses over a given threshold. If there are no or few claims above the threshold, the reinsurer may pass a share of his profits to the direct insurer. The share in the profits is treated as a transfer from the reinsurer to the policy holder.

3. Social insurance schemes

17.13 A social insurance scheme is one where the policyholder is obliged or encouraged to insure against certain contingencies by the intervention of a third party. For example, government may oblige all employees to participate in a social security scheme; employers may make it a condition of employment that employees participate in an insurance scheme specified by the employer; an employer may encourage employees to join a scheme by making contributions on behalf of the employee; or a

trade union may arrange advantageous insurance cover available only to the members of the trade union. Contributions to social insurance schemes are usually paid by, or on behalf of employees, though under certain conditions non-employed or self-employed persons may also be covered.

- 17.14 A social insurance scheme is an insurance scheme where the following two conditions are satisfied:
 - (a) the benefits received are conditional on participation in the scheme and constitute social benefits as this term is used in the SNA; and
 - (b) at least one of the three conditions following is met:
 - (i) Participation in the scheme is obligatory either by law for a specified category of worker, whether employed or nonemployed, or under the terms and conditions of employment of an employee, or group of employees;
 - (ii) The scheme is a collective one operated for the benefit of a designated group of workers, whether employed or nonemployed, participation being restricted to members of that group;
 - (iii) An employer makes a contribution (actual or imputed) to the scheme on behalf of an employee, whether or not the employee also makes a contribution.
- 17.15 Those participating in social insurance schemes make contributions to the schemes and receive benefits. Contributions and benefits are defined in similar ways to insurance premiums and claims. A social insurance contribution is the amount payable to a social insurance scheme in order to be entitled to receive the social benefits covered by the scheme. Social insurance benefits are social benefits payable to a beneficiary because the beneficiary participates in a social insurance scheme.

Social benefits

- 17.16 Social benefits are similar whether paid for by a social insurance scheme or by social assistance. The crucial difference between social assistance and social insurance is that social insurance benefits are only payable if the beneficiary participates in the social insurance scheme paying the benefits. Participation is demonstrated by contributions being paid to the scheme by, or on behalf of, the beneficiary. Social assistance is not dependent on contribution but may be means-tested. There is a section discussing the difference between social insurance and social assistance at greater length in chapter 8.
- 17.17 No matter how they are provided, the characteristics of all social benefits can be described in the following general terms. Social benefits may be provided in cash or in kind. They become payable when certain events occur, or certain conditions exist, that may adversely affect the welfare of the households concerned either by imposing additional demands on their resources or reducing their incomes. There are a number of circumstances in which social benefits may be payable:
 - The beneficiaries, or their dependants, require medical, dental or other treatment, or hospital, convalescent or long-term care, as a result of sickness, injuries, maternity, chronic invalidity, old age, etc. The social benefits are usually provided in kind in the form of treatments or care provided free or at prices that are not economically significant, or reimbursing expenditures made by households. Social benefits in cash may also be payable to beneficiaries needing health care;
 - (b) The beneficiaries have to support dependants of various kinds: spouses, children, elderly relatives, invalids, etc. The social benefits are usually paid in cash in the form of regular dependants' or family allowances;
 - (c) The beneficiaries suffer a reduction in income as a result of not being able to work, or to work full-time. The social benefits are usually paid in cash regularly for the duration of the condition. In some

instances a lump sum may be provided additionally or instead of the regular payment. People may be prevented from working because of

- (i) voluntary or compulsory retirement:
- (ii) involuntary unemployment, including temporary lay-offs and short-time working;
- (iii) sickness, accidental injury, the birth of a child, etc., that prevents a person from working, or working full time;
- (d) The beneficiaries receive payments to compensate for suffering a reduction in income because of the death of the main income earner.
- (e) The beneficiaries are provided with housing either free or at prices that are not economically significant or by reimbursing expenditure made by households. These are social benefits in kind.
- (f) The beneficiaries are provided with allowances to cover education expenses incurred on behalf of themselves or their dependants. Occasionally education services may be provided in kind.
- 17.18 The above are typical circumstances in which social benefits are payable. However, the list is illustrative rather than exhaustive. It is possible, for example, that under some social insurance schemes other benefits may be payable. Conversely, by no means all schemes provide benefits in all the circumstances listed above. In practice, the scope of social insurance schemes is liable to vary significantly from country to country, or from scheme to scheme within the same country.
- 17.19 It is convenient in this part of the chapter to exclude pensions from social insurance benefits. These are dealt with in a separate part of this chapter. Most of the payments described under (c) above are therefore excluded though such payments as severance payments or invalidity benefits are included.

Individual policies qualifying as social insurance

- 17.20 Many social insurance schemes are organized collectively for groups of workers so that those participating do not have to take out individual insurance policies in their own names. In such cases, there is no difficulty distinguishing social insurance from insurance taken out on a personal basis. However, some social insurance schemes may permit, or even require, participants to take out policies in their own names. The determinants for the insurance to count as a social insurance policy are that the benefits must be of the social benefit type and an employer makes an actual or imputed contribution to the scheme on behalf of an employee.
- 17.21 The premiums payable, and claims receivable, under individual policies taken out under a social insurance scheme are recorded as social contributions and social insurance benefits. Most individual policies that qualify as social insurance schemes are likely to be for pension provision but it is possible that they may cover other eventualities, for example to provide income if the policy holder is unable to work for a prolonged period because of ill-health.
- 17.22 Individual insurance policies that do not qualify as social insurance are described as individual insurance not qualifying as social insurance, or in short as other insurance.

4. The units involved

17.23 The institutional units involved in other insurance pre-eminently individual are In principle it is insurance corporations. possible for another type of enterprise to carry out insurance as a non-principal activity, but usually the legal regulations surrounding the conduct of insurance mean that a separate set of accounts covering all aspects of the insurance activity must be kept and thus in the System a separate institutional unit, classified to the insurance corporations and pension funds subsector, is identifiable. Sometimes government may conduct other insurance activities, but again it is likely that a separate unit can be identified. Having noted that exceptionally other sectors may be involved, in what follows it is assumed that all other insurance is carried

out by insurance corporations, either resident or non-resident.

- 17.24 Unlike direct insurance and reinsurance, social insurance schemes can be operated in a number of ways. They are usually organized either
 - a. by government via social security schemes,
 - b. by employers on behalf of their employees and their dependants or
 - c. by others, for example a trade union, on behalf of a specified group.

The last two cases are here called employmentrelated social insurance schemes other than social security. They may be arranged with an insurance corporation as a group policy or series of policies or they may be managed by an insurance corporation in return for a fee. Alternatively, the schemes may be managed by an employer directly on his own behalf.

Multi-employer schemes

17.25 An insurance corporation may, for a fee, agree not only to manage a pension fund but to take on the risks associated with it. This is done in the context of performing this service for a number of schemes collectively under what is called a multi-employer scheme. Under many such schemes, the insurance corporation takes over the responsibility of managing the funds at its disposal so as to make sufficient funds available to meet pension liabilities and to make a surplus it can retain. If it fails to make sufficient funds available for the pension entitlements, it is then the responsibility of this firm and not the original employers, to make good the difference from its own resources.

Coverage of social security

- 17.26 Governments may make separate provision for social insurance for their own employees quite distinct from social security. If this occurs, then these social insurance schemes are treated as if they are private schemes.
- 17.27 In some countries, government may merge the provision of social benefits for its own employees arising from its responsibility as employer with the social security schemes while

- the provision of social benefits for other employees arising from their employers' responsibilities are not included in social security.
- 17.28 In some countries, though, social security may be used by government as a vehicle to absorb all responsibilities of social insurance schemes

throughout the economy. In this case, government is in effect acting as a multiemployer scheme and absolving the individual employers of the risk of excessive pension liabilities and individual employees of the risk of their employer not being able to meet their pension commitments.

B. The measurement of output of direct insurance

- 17.29 As noted at the outset of this part, the process of recording insurance in the System involves decomposing actual transactions and imputing others. This section describes how the various transactions to be recorded in the System are derived. The transactions to be recorded are the same whether the type of insurance is an other individual policy or relates to the operation of a social insurance scheme. For this reason in this section the terms premium and claims should be understood to include contributions and benefits respectively.
- 17.30 Once the various transactions have been described in this section, their recording is described in section D after the methods of measuring and recording reinsurance have been discussed.

1. Property income attributed to policyholders/beneficiaries

17.31 For both life and non-life insurance, the total amount of claims payable in a given period often exceeds the premiums receivable. The insurance corporation can accept this because the contingencies covered by the policies do not occur, even for the whole population covered, at the same time as the premiums are paid. Premiums are usually paid regularly, often at the start of an insurance period, whereas claims fall due later, in the case of life insurance often many years later. In the time between the premium being paid and the claim being payable, the sum involved is at the disposal of the insurance corporation to invest and earn income from it. The income thus earned allows the insurance corporations to charge lower premiums than would be the case otherwise. An adequate measure of the service provided must take account of the size of this income as

well as the relative size of premiums and claims.

- 17.32 The income concerned comes from the investment of the provisions of the insurance corporations, which represent liabilities towards the policyholders. For non-life insurance, even though a premium may be payable at the start of a period of cover, the premiums are only earned on a continuous basis as the period passes. At any point before the end of the cover, the insurance corporation holds an amount due to the policy holder relating to services and possible claims to be provided in the future. This is a form of credit extended by the policyholder to the insurance corporation described as unearned premiums. Similarly, although claims become due for payment by the insurance corporation when the contingency specified in the policy eventuates, they may not be actually payable until some time later, often because of negotiation about the amounts due. This is another similar form of credit, described as reserves against claims outstanding.
- 17.33 Similar provisions exist for life insurance but in addition there are two other elements of insurance provisions, actuarial reserves for life insurance and reserves for with-profit insurance. They represent amounts set aside for payments of claims and benefits in future that exceed the receipts of premiums and contributions received to the current date. Usually the provisions are invested in financial assets and the income is in the form of investment income (interest and dividends). Sometimes, however, they may be used to generate net operating surplus either in a separate establishment or as a secondary activity. The most common example is from real estate.

- 17.34 It is common with life insurance policies for amounts to be explicitly attributed by the insurance corporation to the policyholders in each year. These sums are often described as bonuses. The sums involved are not actually paid to the policy holders but the liabilities of the insurance corporation towards the policy holders increase by this amount. This amount is shown as property income attributed to the policy holders. The fact that some of it may derive from holding gains does not change this designation; as far as the policy holders are concerned it is the return for making the financial asset available to the insurance corporation. In addition, all the income from the investment of non-life provisions, and any excess of income from the investment of life provisions (excluding holding gains) over any amounts explicitly attributed to the policy holders, is shown as property income attributed to policyholders, regardless of the source of the income.
- 17.35 All property income attributed to policy holders, whether explicitly by the insurance corporation or implicitly within the System, is shown as premium supplements or contribution supplements.
- 17.36 For direct non-life insurance, the property income attributed to the policy holders should, in principle, be made according to the proportion of provisions attributed to the different classes of insurance and policyholders. In practice, the usual method is to distribute the property income in proportion to the actual premiums payable. For direct life insurance and social insurance schemes, the property income is attributed to households (possibly including some non-resident households).

2. Non-life insurance claims

17.37 The level of claims made on non-life insurance policies varies from year to year and there may be exceptional events that cause a particularly high level of claims. However, the insurance company sets the level of premiums on the basis of past experience of claims and for this reason, the formula for the calculation of output should use not actual claims but an adjusted figure. The term "adjusted" is used here to confirm that it is not observed claims, indeed the production of insurance services does not occur when the risk occurs. The concept of insurance service is the service of covering for the risk. As such, its

measurement should not be affected by the volatility of the occurrence of the risk. Neither the volume nor the price of insurance services is directly affected by the volatility of claims.

- 17.38 The figure for adjusted claims may be derived statistically based on previous experience of the level of claims. Alternatively, an accounting approach may be adopted whereby figures from the accounts of insurance corporations including equalisation provisions may be used. The accounting allowance for equalisation provision is also an adjustment to reflect the variations in claims from one year to another. Whichever method is used, therefore, the adjusted claim figure approximates the expected level of claims.
- 17.39 Immediately after a disaster out of line with previous experience, the level of expected claims will be higher, consistent with the observation that after a disaster, premiums rise. There is never a reason to adjust the level of adjusted claims retrospectively in the light of the exceptional disaster; it is only future expectations that reflect the impact of this disaster.

3. Defining insurance output

- 17.40 The output of the insurance corporation represents the service provided to the policy holders. The output of direct non-life insurance is based on the principle of adding premiums and premium supplements and deducting adjusted claims incurred.
- 17.41 If a statistical approach is being used, the formula to calculate output takes the following form:

Actual premiums earned;

Plus premium supplement,

Less adjusted claims;

where adjusted claims are estimated from past experience. In such a case, conceptually premium supplements should also be estimated on the basis of past experience. However, since premium supplements are less volatile than claims, in practice no such adjustment may be necessary. If a statistical basis is to be used for estimating output, it is preferable to use

information broken down by "line of business", that is for motor insurance, buildings insurance, etc. separately.

17.42 Alternatively, an accounting approach may be used whereby output is calculated as:

Actual premiums earned;

Plus premium supplements;

Less adjusted claims;

where adjusted claims are determined by using claims due plus the changes in equalisation provisions and, if necessary, changes to own funds.

- 17.43 If the necessary accounting data are not available and the historical statistical data are not sufficient to allow reasonable average estimates of output to be made, the output of non-life insurance may be estimated as the sum of costs (including intermediate costs, labour and capital costs) plus an allowance for "normal profit".
- 17.44 In all these formulae, premiums may be replaced by contributions and claims by benefits.
- 17.45 The output of direct life insurance is calculated separately as:
 - (a) Actual premiums or contributions earned;
 - (b) *Plus* premium or contribution supplements;
 - (c) Less claims or benefits due;
 - (d) Less increases (plus decreases) in actuarial reserves and reserves for with-profits insurance.
- 17.46 If adequate data are not available for the calculation of life insurance according to this formula, an approach based on the sum of costs, similar to that described for non-life insurance may be used.

4. Net premiums/contributions and consumption of insurance services

- 17.47 The actual premiums or contributions payable and the premium supplements or contribution supplements are shown in the System divided between two types of transactions. The first is the value of the output of insurance, which is shown as either consumption or export of insurance services. The second is net premiums or contributions earned by the insurance corporations. Net premiums are defined as actual premiums plus premium supplements less the insurance service charge payable by the policy holders. Because of the way in which the value of the service output is defined, net premiums for non-life insurance are equal in total to adjusted, and not actual, claims. Any variation between adjusted and actual claims represents a transfer between the policy holders and the insurance corporation. Over time, a transfer in one direction is offset by one in the other.
- 17.48 Insurance services are consumed by those sectors (and the rest of the world) that pay premiums and contributions. Estimates of the value of consumption by sector are usually made by allocating the total value of the service in proportion to the actual premiums or contributions payable. Estimates of net premiums or contributions are then made by deducting the consumption of services from the total actual premiums or contributions payable plus the value of the premium or contribution supplements. (Because premium contribution supplements are also allocated in proportion to actual premiums, the net premiums and contributions are allocated in the same proportions as the actual premiums and contributions.)

5. Recording claims

- 17.49 The time of recording claims incurred is in the period in which the event to which the claim relates took place. This principle is applied even when, in the case of disputed claims, the settlement may take place years after the event concerned.
- 17.50 Claims are normally recorded as current transfers payable by the insurance corporation to the policy holder. However, when the level of claims is exceptionally large (as compared

- with the experience of the insurance corporation of the level of claims settled), the excess may be recorded as a capital transfer from the insurer to the policy holder.
- 17.51 Because the formula for output uses adjusted claims and not actual claims, only when the actual claims happen to be the same level as expected claims will net premiums and claims be equal in a given period. They should however be approximately equal over a period of years excluding a year in which a disaster is recorded.

Insurance services provided to and from the rest of the world

17.52 Resident insurance corporations frequently provide insurance cover to households and enterprises in the rest of the world, and resident households and enterprises may purchase cover from insurance corporations in the rest of the world. The property income attributed by resident insurance corporations to policyholders includes an allocation to policyholders in the

- rest of the world. These non-resident policyholders then also pay premium supplements to the resident insurance corporation. This information should be available for resident insurers and should be included in the rest of the world account.
- 17.53 Similar considerations also apply to the treatment of resident enterprises and households taking out policies with non-resident insurers. They receive imputed property income from abroad and pay premiums and supplements to abroad. Estimation of the size of these flows is more difficult, especially when there is no resident insurer of the same type against which to make comparisons. However, very often the country providing the service will be known and it may be possible to use counterpart data to make estimates for the national economy. The level of transactions by residents should be known and the ratio of premium supplements to actual premiums in the economy providing the services could be used to estimate the property income receivable and premium supplements payable.

C. The measurement of reinsurance

- 17.54 Before discussing how the various elements contributing to the measurement of output of insurance are recorded in the System, it is necessary to describe how reinsurance is measured and recorded.
- 17.55 The transactions between the direct insurer and the policy holder are measured as described in the previous section without any reference to the transactions between the direct insurer and the reinsurer. The transactions between the direct insurer and the reinsurer are recorded as an entirely separate set of transactions and no consolidation takes place between the transactions of the direct insurer as issuer of policies to its clients on the one hand and the holder of a policy with the reinsurer on the other.
- 17.56 The direct policy holder does not know, or need to know, whether the direct insurer involves a reinsurer to protect it against loss on the policy. The direct insurer receives actual premiums from its policy holders. Some of these are

- ceded to a reinsurer. The premiums are shown as being first payable to the direct insurer and then some are payable to the reinsurer. This non-consolidation is sometimes referred to as gross recording on the part of the direct insurer. The alternative (net recording) would be to show part of the direct policy holders premiums being paid to the direct insurer and part to the reinsurer but this option is not recommended either in commercial accounting or in the System.
- 17.57 The actual premium payable by the direct insurer to the reinsurer is used by the reinsurer to earn investment income. This is treated as property income payable to the direct insurer and returned to the reinsurer as a premium supplement. Thus a direct insurer pays property income to its policy holders based on the whole of the premiums earned (or by approximation payable) to the direct insurer but receives as an offset property income from the reinsurer corresponding to the amount of the premiums it has ceded to the reinsurer.

- 17.58 The formulae to calculate the output of reinsurance services is exactly analogous to those for direct insurance. However, because the primary motivation of reinsurance is to limit the direct insurer's exposure to risk, a reinsurer deals with exceptionally large claims as a matter of normal business. For this reason, and because the market for reinsurance is concentrated in relatively few large firms world-wide, it is less likely that the reinsurer will experience an unexpectedly large loss than a direct insurer does, especially in the case of excess of loss reinsurance.
- 17.59 Although both life and non-life direct insurance may be subject to reinsurance, all reinsurance policies are treated in a similar way to non-life insurance. As noted earlier, they may be some payments peculiar to reinsurance. These are commissions payable to the direct insurer under proportionate reinsurance and profit sharing in

excess of loss reinsurance. Once these are taken into account the output of reinsurance can be calculated as:

- (a) Total actual premiums earned less commissions payable;
- (b) Plus premium supplements;
- (c) Less both adjusted claims due and profit sharing.
- 17.60 The whole of the output of the reinsurer represents intermediate consumption of the direct insurer holding the reinsurance policy. As noted above, many reinsurance policies are between insurance corporations resident in different economies. Thus the value of the output in these cases represent imports or exports.

D. The recording of non-life insurance and social insurance other than pensions

- 17.61 As explained above, the total of premiums (contributions) earned and the supplements payable by policyholders are decomposed into two elements, the part corresponding to the consumption of the service provided by the insurance corporations and net premiums. It follows that the entries for insurance in the accounts of the System consist of two types of transactions. The first set corresponds to the production and consumption of insurance services. The second set corresponds to the payments of net premiums (contributions) and claims (benefits). As explained above, there is an approximate equality between premiums and claims over the long run. These transactions are thus essentially redistributive and are mainly recorded in the secondary distribution of income account as current transfers.
- 17.62 Because the interaction of these two sets of transactions can seem complicated, the entries associated with the different types of insurance schemes in the System are described in turn below in ascending order of complexity. The first two cases discussed are those of the social security schemes of government and employers' unfunded schemes where no insurance services are separately recorded. The next case considered is the most familiar case of

- insurance, other non-life insurance, which covers accident, fire, vehicle, etc., insurance. The description of other life insurance follows. Lastly, the case of social insurance under private funded schemes, where more elaborate recording of contributions and benefits is required, is described.
- 17.63 In each description, the type of transactions and the sectors involved are specified. The only exception is for imported insurance services. Not to over-complicate the exposition, the possibility of transactions with non-resident insurers is not included explicitly below. It should be remembered however, that every transaction with an insurance corporation could in principle be with a non-resident rather than resident institutional unit, in which case the transaction should be recorded in the accounts for the rest of the world rather than with the insurance corporation and pension fund subsector. For reinsurance it is especially likely that transactions with the rest of the world are involved

1. Social security schemes of government (excluding pensions)

17.64 Most governments operate a social security scheme where employees, the self-employed and occasionally non-employed persons make contributions. However, benefits may also be provided to some people who have not contributed to the scheme but are designated as beneficiaries of a relative who has contributed. Contributions to the social security scheme may be made directly by employees, the self- and non-employed or on behalf of employees by employers. The value of these latter payments are treated in the System as part of compensation of employees, that is, payable by employers to employees and then by employees to government in addition to the contributions they make on their own behalf as employees.

17.65 In the System flows are recorded as follows.

- (a) Employers' social security contributions are shown as payable by the sector in which the employer is located and receivable by households. The sector of the employer may be any of non-financial corporations, financial corporations, general government (as an employer), employer households, NPISHs or the rest of the world (when a resident works for a non-resident institutional unit). resident employers the payables are shown in the generation of income account; payables by non-resident employers are shown in the primary distribution of income account for the rest of the world. Receivables by resident households are shown in the allocation of primary income account and by non-resident households in the primary distribution of income account for the rest of the world.
- (b) In the secondary distribution of income account, the sum of employers' social security contributions and social security contributions by households in their capacities as employees is shown as payable by households and receivable by government. Further, social security benefits in cash payable to households are shown as payable by government (or the rest of the world if from a foreign government) and receivable by households.

17.66 An example of these flows is shown in table 17.1.

2. Unfunded social insurance schemes operated by employers

- 17.67 Not all employers offer social benefits other than pensions and even those that do may fund them on a pay as you go basis. In this case, the flows are similar to those above but all flows are between households and employers.
- 17.68 In the System, an employer operating an unfunded scheme is regarded as making an imputed social contribution to the scheme on behalf of the employees. In practice, the value of contributions is usually set equal in value to the benefits payable in the period under consideration (plus the cost of operating the scheme described in the following paragraph). The imputed contribution forms part of the compensation of employees and is also shown as being payable by the employees to the scheme together with any actual payments by the employees. However, it is not uncommon for unfunded schemes to be non-contributory for the employees.
- 17.69 Even if a scheme is unfunded, there are costs involved in administering it. Output equal to the sum of these costs should be imputed as being paid for by the beneficiaries from an imputed element of contributions. The imputed contribution to employees should include these costs as well as the value of the benefits received by employees. The amount is then recorded in the use of income account as a purchase of a service by the employees from the employer.
- 17.70 There are two transactions recorded for the production and consumption of the services provided by the employer. Because the scheme is unfunded, there are no property income flows and no contribution supplements to be recorded. There are two sets of redistributive transactions recorded.
- 17.71 The production and consumption transactions are as follows.
 - 1. Output of services is imputed in the production account of the employer and the value of the output forms part of the imputed employers' contributions to

- social insurance incorporated in compensation of employees.
- Consumption of the service is recorded in the use of income account for resident households or as exports for non-resident households.
- 17.72 The redistributive transactions are as follows.
 - (a) Employers' imputed contributions to unfunded social insurance schemes are shown as a payable by the sector in which the employer is located in the generation of income account and a receivable by households in the allocation of primary income account.
 - (b) In the secondary distribution of income account, employers' imputed contributions and actual contributions by employees are shown as payable by households and receivable by the employer. Further, benefits payable to households by the employer are shown as payable by the employer and receivable by households.
- 17.73 An example of these flows is shown in table 17.2.

3. Individual non-life insurance other than policies qualifying as social insurance

17.74 In order to record all the transactions required, it is necessary to consider the transactions relating to direct insurance in a first step and then consider the further transactions necessary for reinsurance. For practical purposes, it is possible consider direct insurance to transactions as taking place between units in the insurance and pension fund sub-sector and units in any other sector and reinsurance transactions as taking place within the insurance and pension funds sub-sector, remembering that nonresidents may have policies with resident corporations insurance and insurance corporations may issue policies to nonresidents. Altogether six pairs of transactions need to be recorded in respect of other individual non-life insurance; two pairs relating to the measurement of the production and consumption of the insurance service, three pairs relating to redistribution and one in the financial account. Under certain circumstances, a seventh transaction relating to redistribution

may be recorded in the capital account. The value of the output of the activity, the property income to be attributed to the policyholders and the value of the service charge are calculated specifically for other non-life insurance in the manner described above for both direct insurance and reinsurance.

- 17.75 The production and consumption transactions are as follows:
 - (a) Since all such activity by resident institutional units is undertaken by insurance corporations, the output is recorded in the production account of insurance corporations;
 - The service may be consumed by any of the sectors of the economy or by the rest of the world; the value of the service is payable to insurance corporations. Payments by non-financial corporations, financial corporations, general government non-profit institutions constitute intermediate consumption, recorded in their production account. Insurance clearly associated with the productive activity of an unincorporated enterprise is also recorded as intermediate consumption in the production account of households. Other insurance payments by households are part of final consumption expenditure, recorded in the use of income account. Payments by the rest of the world are recorded as exports in the external account of goods and services.
 - (c) Reinsurance services are also produced by insurance corporations (or is imported from abroad) and are supplied to insurance corporations, either resident or nonresident where their purchase forms part of intermediate consumption or exports, respectively.
- 17.76 The redistributive transactions cover property income attributed to policyholders in respect of other non-life insurance, net non-life insurance premiums, and insurance claims:
 - (a) Property income attributed to policyholders in respect of other non-life insurance is recorded as payable by insurance corporations. It is recorded as receivable by all sectors and the rest of the

- world. Both payables and receivables are recorded in the allocation of primary income account.
- (b) Property income receivable by reinsurance policy holders is payable by insurance corporations and receivable by similar institutions either resident or non-resident.
- (b) Net non-life insurance premiums are calculated as premiums earned plus premium supplements (equal to the property income attributed to policyholders) less the value of the services consumed. These net premiums are payable by all sectors of the economy or the rest of the world and receivable by insurance corporations. (This applies to both direct insurance and reinsurance.)
- (c) Insurance claims due are payable by insurance corporations and receivable by all sectors of the economy and the rest of the world. Reinsurance claims are payable by insurance corporations and receivable by insurance corporations, either resident or non-resident. Both net premiums and claims are recorded in the secondary distribution of income account.
- (d) If insurance claims are exceptionally large in a period as a result of a disaster, natural or man-made, the claims recorded in the secondary distribution of income account may be reduced to their normal or expected levels and the remainder recorded as a capital transfer payable by insurance corporations and receivable by all sectors of the economy and the rest of the world. (This applies to both direct insurance and reinsurance, bearing in mind that a corporation issuing reinsurance policies probably has a larger tolerance for expected claims than a direct insurer.)
- 17.77 Net non-life insurance premiums should be recorded on the basis of the amounts due to obtain cover in the period of account, not the amounts due to be paid in the period. Insurance claims should be recorded on the basis of the amounts due at the date of the event concerned occurred. An entry in the financial account records any difference between premiums payable and premiums earned and claims due and claims payable:

- (a) By convention, unearned premiums and provisions against outstanding claims is shown as a change in liabilities of insurance corporation (with a negative sign if necessary) and a change in assets of all sectors and the rest of the world.
- (c) Changes in equalisation provisions are recorded in the other changes in assets account.
- Similar recordings take place for both direct insurance and reinsurance.
- 17.78 An example of these flows is shown in table 17.3.
- 4. Individual life insurance other than policies qualifying as social insurance and annuities
- 17.79 Note that reinsurance transactions are recorded in the accounts for direct insurance even if the reinsurance policies cover claims on life policies.
- 17.80 Individual life insurance transactions take place only between insurance corporations and households, resident and non-resident. The production of the insurance services is matched by the value of the services consumed by households as part of final consumption expenditure and exports. The property income attributed to insurance policyholders is treated as premium supplements. However, premiums and claims are not shown separately in the case of other life insurance. Rather they constitute components of a net transaction recorded in the financial account, the financial asset involved being the net equity of households in life insurance reserves.
- 17.81 Four transactions are recorded in the accounts; one each relating to production and consumption of the insurance service, one showing the attribution of property income to the property holders and one showing the change in the net equity of households in life insurance reserves:
 - (a) The output of the life insurance activity is recorded in the production account for the insurance corporations.

- (b) The value of the services consumed is recorded as final consumption expenditure payable by households in the use of disposable income account or as payable by the rest of the world (exports to nonresident households).
- (c) Property income attributed to insurance policyholders in respect of other life insurance is recorded in the allocation of primary income account. Bonus declared in connection with life policies are treated as property income even if they are exceed the property income earned by the institution declaring the bonus. The property income is recorded as payable by insurance corporations and receivable by resident households or non-resident households in the rest of the world.
- (d) In the financial account, the item life insurance and annuities entitlements is shown as a change in assets of households and the rest of the world and a change in liabilities of insurance corporations. It is equal to actual premiums plus premium supplements (equal to the property income attributed to policyholders) less the value of the services consumed and less claims incurred.
- 17.82 An example of these flows is shown in table

5. Funded social insurance other than pensions

17.83 As noted above, funded schemes for benefits other than pensions are not very common. They may, however, exist in two circumstances. The first is when an employer has a fund for such benefits and accumulates any under spend in one year to pay for possible overspends in future years. Alternatively, an employer may realise that the commitments to make payments in future are such that it is prudent to build reserves to be able to make such payments. An example of such a scheme might be one that provides health cover to present and past employees. Unlike in the case of pensions, no estimates are included in the System of possible future claims on social insurance benefits other than pensions and liabilities are recorded only when and to the extent that they exist in the employer's accounts.

- 17.84 Funded social insurance covering benefits other than pensions may be carried out by insurance corporations or by employers on behalf of their The output of this activity is employees. measured in the same way as the output of nonlife insurance but the matching consumption of the services is payable only by the households of the beneficiaries. These will be resident households except where a resident producer is liable to pay benefits to a present or former employee who is a non-resident or who has a non-resident family member entitled to the benefits. The property income attributed to the beneficiaries of the social insurance schemes can only be receivable by the same households.
- 17.85 All contributions to the schemes are recorded as payable by the employee. These contributions include that part paid by the employer as part of compensation of employees in the generation of income account as well as contributions paid directly by the employee funded from wages and salaries. Further, the employee receives property income attributed to policyholders in respect of these contributions and this is treated, in total, as contribution supplements. Two items of contributions appear in the secondary distribution of income account. The first, the employers' actual social contributions is exactly equal in value to the amount receivable by households from the employer in the generation of income account. The second item, called employees' social contributions includes the direct payment by the employees plus the contribution supplements less the service charge payable to the social insurance schemes.
- 17.86 Seven types of transactions must be recorded, one each relating to production and consumption of the insurance service, three relating to contributions and benefits, one to the property income attributable to policyholders and one to an adjustment in the financial account:
 - (a) The activity by resident units is undertaken by insurance corporations or by an employer; the output is recorded in the production account of the insurance corporations or in the sector of the employer as appropriate.
 - (b) Employers' actual social contributions to employment-related social insurance schemes are shown as payable by the

- sector in which the employer is located in the generation of income account and receivable by households in the allocation of primary income account;
- (c) Property income attributed to policyholders (beneficiaries) in respect of these schemes is payable by insurance corporations and employers, and receivable by employee households. Both payables and receivables are recorded in the allocation of primary income account;
- (d) Net social contributions are shown in the secondary distribution of income account as payable by households and receivable by insurance corporations or the sector of the employer as appropriate.
- (e) Employment-related social benefits other than pensions are also shown in the secondary distribution of income account as payable by insurance corporations or the sector of the employer and receivable by households;
- (f) The value of the service is payable by households as part of final consumption expenditure, and is recorded in the use of income account, except for non-resident employee households where it is payable by the rest of the world;
- (g) The entry in the financial account, provisions for other social benefits, records any difference between contributions payable and contributions

earned and benefits due and benefits payable. This item is shown as a change (with a negative sign if necessary) in provisions for other social benefits of insurance corporations (or employer sector) and a change in assets of employee households.

17.87 An example of these flows is shown in table 17.5.

6. A note on the tables

- 17.88 Tables 17.1 to 17.5 illustrate the cases described in the immediately preceding sections. The tables are not in strict conformity with tables in chapters 6 to 13. The transactions listed are at a finer level of detail that the norm in order to show exactly what type of payment is being discussed. Further, the columns of the tables do not conform to institutional sectors in the System but illustrate the major agents in each type of insurance scheme under discussion. Where "employer" is shown as one of the agents, in practice the employer may be in any of the institutional sectors or in the rest of the world. (It is unusual but conceptually possible for a household to provide social insurance to employees of an unincorporated enterprise.)
- 17.89 In order to show how these tables integrate with the full set of accounts, table 17.9 shows how the very detailed transactions in these tables as well as those in the pensions part can be aggregated to reach values that run through tables in other chapters.

Table 17.1: Accounts for non-pension benefits paid through social security - uses

ш	c	Δ

	Social security				
	Employer	fund	Households	Other sectors Total economy	
Generation of income account					
Employers' actual social security contributions (non-pension)	15.0			15.0	
Distribution of primary income					
Employers' actual social security contributions (non-pensions)					
Secondary distribution of income account					
Social security contributions (non-pensions)			25.0	25.0	
Employers' actual social security contributions (non-pensions)			15.0	15.0	
Household actual social security contributions (non-pensions)			10.0	10.0	
Social security non-pension benefits		22.0		22.0	

Table 17.2: Accounts for unfunded non-pension social insurance benefits - uses

U	١s	e.

	Employer	Social insurance fund	Households	Other sectors Total economy
Generation of income account Employers' imputed non-pension contributions	9.0			9.0
Distribution of primary income Employers' imputed non-pension contributions				
Secondary distribution of income account				
Household total non-pension contributions Employers' imputed non-pension contributions			9.0 9.0	9.0 9.0
Unfunded non-pension benefits		9.0		9.0

Table 17.3: Accounts for individual non-life insurance - uses

Uses

	Corporations	Insurance corporations	Households	Other sectors	Total economy
Production account	Oorporations	corporations	riouscrioius	Other Sectors	rotal economy
Intermediate consumption	1.0		3.0		4.0
Output					
Distribution of primary income					
Property income attributable to non-life insurance policy holders		6.0			6.0
Secondary distribution of income account					
Net non-life insurance premiums	8.0		31.0	6.0	45.0
Non-life insurance claims			45.0		45.0
Use of income					
Final consumption expenditure			2.0		2.0
Financial account					
Non-life insurance technical provisions			3.0		3.0

Table 17.4: Accounts for individual life insurance - uses

Uses

		Insurance		
	Corporations	corporations	Households	Other sectors Total economy
Production account				
Output				
Distribution of primary income				
Property income attributable to life insurance policy holders		7.0		7.0
Use of income				
Final consumption expenditure			4.0	4.0
Financial account				
Life insurance and annuity entitlements			22.0	22.0

Table 17.1:Accounts for non-pension benefits paid through social security -resources

				Resources
	Soc	ial security		
	Employer	fund	Households	Other sectors Total economy
Generation of income account				_
Employers' actual social security contributions (non-pension)				
Distribution of primary income				
Employers' actual social security contributions (non-pensions)			15.0	15.0
Secondary distribution of income account				
Social security contributions (non-pensions)		25.0		25.0
Employers' actual social security contributions (non-pensions)		15.0		15.0
Household actual social security contributions (non-pensions)		10.0		10.0
Social security non-pension benefits			22.0	22.0

Table 17.2: Accounts for unfunded non-pension social insurance benefits - resources

				Resources
		Social insurance		
	Employer	fund	Households	Other sectors Total economy
Generation of income account Employers' imputed non-pension contributions				
Distribution of primary income Employers' imputed non-pension contributions			9.0	9.0
Secondary distribution of income account				
Household total non-pension contributions		9.0		9.0
Employers' imputed non-pension contributions		9.0		9.0
Unfunded non-pension benefits			9.0	9.0

Table 17.3: Accounts for individual non-life insurance - resources

					Resources
		Insurance			
	Corporations	corporations	Households	Other sectors	Total economy
Production account					
Intermediate consumption					
Output		6.0			6.0
Distribution of primary income					
Property income attributable to non-life insurance policy holders	5.0		1.0		6.0
Secondary distribution of income account					
Net non-life insurance premiums		45.0			45.0
Non-life insurance claims	6.0		35.0	4.0	45.0
Use of income					
Final consumption expenditure					
Financial account					
Non-life insurance technical provisions		3.0			3.0

Table 17.4: Accounts for individual life insurance - resources

				Resources
		Insurance		
	Corporations	corporations	Households	Other sectors Total economy
Production account				
Output		4.0		4.0
Distribution of primary income				
Property income attributable to life insurance policy holders			7.0	7.0
Use of income				
Final consumption expenditure				
Financial account				
Life insurance and annuity entitlements		22.0		22.0

Table 17.5: Accounts for funded non-pension social benefits - uses

Uses Social insurance Employer fund Households Other sectors Total economy Production account Output Generation of income account Employers' actual non-pension contributions 6.0 6.0 Distribution of primary income Employers' actual non-pension contributions Property income 4.0 4.0 Property income payable on non-pension entitlements 4.0 4.0 Secondary distribution of income account Household total non-pension contributions 14.0 14.0 Employers' actual non-pension contributions 6.0 6.0 Household actual non-pension contributions 5.0 5.0 Household non-pension contribution supplements 4.0 4.0 Social insurance scheme service charges -1.0 -1.0 Funded non-pension benefits 13.0 13.0 Use of income Final consumption expenditure 1.0 1.0 Change in non-pension entitlements 1.0 1.0 Saving -6.0 1.0 9.0 -4.0 0.0

Table 17.5: Accounts for funded non-pension social benefits - resources

	-			Resources
	Employer	Social insurance fund	Households	Other sectors Total economy
Production account				
Output		1.0		1.0
Generation of income account				
Employers' actual non-pension contributions				
Distribution of primary income				
Employers' actual non-pension contributions Property income Property income payable on non-pension entitlements		4.0	6.0 4.0	6.0 4.0 4.0
Secondary distribution of income account				
Household total non-pension contributions Employers' actual non-pension contributions Household actual non-pension contributions Household non-pension contribution supplements Social insurance scheme service charges Funded non-pension benefits		14.0 6.0 5.0 4.0 -1.0	13.0	14.0 6.0 5.0 4.0 -1.0 13.0
Use of income				
Final consumption expenditure Change in non-pension entitlements Saving			1.0	1.0

E. Annuities

- 17.90 The simplest case of a life insurance policy is one where a stream of payments is made by the policy holder to the insurance corporation over time in return for a single payment received as a claim at some point in the future. With the simplest form of annuity, the equivalent to the policy holder, called the annuitant, pays a single lump sum to the insurance corporation and in return receives a stream of payments either for a nominated period or for the rest of the annuitant's life (or possibly for the rest of the life of both the annuitant and a nominated other person).
- 17.91 Annuities organised insurance are by corporations and are a means of risk management. The annuitant avoids risk by agreeing to accept a known income stream (known either in absolute terms or subject to a formula, such as being index-linked) in return for parting with a considerable sum. insurance corporation takes the risk of making more from investing the sum than is due to the annuitant. The rates of annuities are determined taking life expectancy into account and longlived annuitants may receive more than their original payment and the income earned on it. Those who die early receive less, possibly considerably less.
- 17.92 The recording of annuities has parallels with the recording of life insurance. The initial payment is recorded in the financial account as an actual premium, initially all unearned. At the same time, the insurance corporation incurs a liability towards the annuitant based on the terms of the contract. Some annuities are for a fixed term, so the calculation of the liability is relatively straightforward. More generally, annuities are for life, in which case actuarial assumptions are made by the insurance corporation about how long the annuitant is expected to live. The liability of the insurance corporation towards the annuitant in most cases will exceed the actual premium because of the expected revenue to be generated by the insurance corporation by investing the funds.
- 17.93 The liability to the annuitant at the start of the year, is equal to the present value of the future

- stream of payments to be made. During the year, the value increases because the future stream is one element shorter and so one future discount factor is applied to calculate the revised present value. The increase in value is shown as property income payable to the annuitant. This amount is determined by applying an appropriate discount factor to the present value at the start of the year and is not related to the amount of property income earned by the insurance corporation in the year.
- 17.94 At the same time, the amount initially paid for the annuity decreases. The decline in the value of the annuity represents the drawdown of the year's payment less the property income payable. The decline in value in fact shows the amount of premium earned in that year. (These transactions can be seen to be analogous to the transactions recorded for the decline in value of fixed capital or the payments made under a financial lease.)
- 17.95 The drawdown of the value plus the property income will however be slightly greater than the payment due under the annuity since it will also include a service charge payable to the insurance corporation through out the life of the annuity. As with life insurance, the value of output is then determined as

premiums earned

plus premium supplements (the property income attributable to the unwinding of the discount factor)

less claims incurred (the agreed amount payable).

Because of the way premium earned is linked to the decline in liability of the insurance corporation, there is no need in this case to have an adjustment terms for the change in actuarial reserves.

17.96 Each year the annuitant receives the sum agreed in the terms of the annuity agreement. The liability of the insurance corporation decreases. For an individual annuity, the liability changes

from year to year also as the actuarial assumptions about that person change. Premiums are earned in proportion to the change in the liability. In this way, the initial premium is earned gradually over the entire period that the annuity is operational. The final element of premium earned is exactly enough to exhaust the initial premium,. In the case where an annuitant dies unexpectedly early, there will be a large earned premium in that year. If an annuitant lives longer than expected, the rate at which premiums are earned will have slowed down as the liability towards that annuitant declines more slowly. (In practice it is probable that the calculations are undertaken for cohorts of people with similar actuarial expectations rather than individually.)

In addition, there is property income earned by the insurance corporation and this is treated as payable to the annuitant and can be referred to as premium supplements

- 17.97 The entries for annuities for all annuitants taken together are as follows:
 - a. The initial premium is recorded in the financial account at the time it is made as the acquisition of a financial asset by the

insurance corporation. At the same time a liability of the insurance corporation towards households is established equal to the present value of the claims to be made under the annuity.

- b. Each year property income earned by the insurance corporation on the unearned premium is shown as property income payable to the annuitant by the insurance corporation in the allocation of primary income account.
- c. In the use of income account, the service charge due for that year is shown as payable by the annuitant in the use of income account and receivable by the insurance corporation in the production account. The value is equal to the value of the premium earned in that year, plus the property income received in the previous step less the claim incurred for that year.
- d. In the financial account, the insurance corporation's liability to the annuitant decreases by the amount of premium earned that year.

Part 2 The treatment of loan guarantees in the System

F. Types of guarantees

- 17.98 A loan guarantee is normally an arrangement whereby one party, the guarantor, undertakes to a lender that if a borrower defaults, the guarantor will make good the loss the lender would otherwise suffer. Often a fee is payable for the provision of a guarantee though the form of this varies. Sometimes the guarantor will acquire some rights over the defaulting borrower.
- 17.99 Guarantees have a significant impact on the behaviour of economic agents, both by influencing their decisions on production, income, investment or saving and by modifying the lending and borrowing conditions on financial markets. Some borrowers would have no access to loans in the absence of guarantees, while others would not benefit from comparatively low interest rates. Guarantees are particularly significant for the general government sector and for the public sector as government activities are often linked with the issuance or activation of guarantees.
- 17.100Three classes of guarantees are recognized. These apply only to guarantees provided in the case of loans. No special treatment is proposed for guarantees in the form of manufacturers' warrantees or other form of guarantee. (The cost of replacing defective merchandise is an intermediate cost of the manufacturer.)
- 17.101The first of these is composed of those guarantees provided by means of a financial derivative, such as a credit default swap. These derivatives are actively traded on financial markets. The derivative is based on the risk of default of a reference instrument and so not actually linked to an individual loan or bond. Incorporating the transactions connected with establishing this sort of financial derivative is discussed in chapter 11.
- 17.102The second class of guarantees, standardized guarantees, is composed of the sorts of guarantees that are issued in large numbers,

- usually for fairly small amounts, along identical lines. There are three parties involved in these arrangements, the borrower (debtor), the lender (creditor) and the guarantor. borrower or lender may contract with the guarantor to repay the lender if the borrower defaults. The classic examples are export credit guarantees and student loan guarantees. Here, although it is not possible to establish the likelihood of any one loan defaulting, it is not only possible but standard practice to estimate how many out of a batch of similar loans will default. If the guarantor is working on purely commercial lines, he will expect all the fees paid, plus the property income earned on the fees, to cover the expected defaults. This is exactly the same paradigm as operates for nonlife insurance and a similar treatment is adopted for these guarantees, described as "standardized guarantees". This involves including transactions and balance sheet items parallel to those for non-life insurance, including the generation of output and payments of a fee supplement and a service fee by those taking out the guarantees.
- 17.103The third class of guarantees, described as oneoff guarantees, consists of those where the loan or the security is so particular that it is not possible for the degree of risk associated with the loan to be calculated with any degree of accuracy. In most cases, the granting of a oneoff guarantee is considered a contingency and is not recorded as a financial asset/liability. (As an exception, one-off guarantees granted by governments to corporations in certain welldefined financially distressed situations and with a very high likelihood to be called are treated as if these guarantees are called when the financial distress is recognised.) If a fee is charged, this is recorded as a payment for a service at the time of payment. If a call is made under a guarantee, a capital transfer is recorded from the guarantor to the guarantee holder at the time of default or, in cases where the guarantor obtains an effective claim on the guarantee

- holder, a financial transaction (including increases in equity participation).
- 17.104 Standardised guarantees are to be distinguished from one-off guarantees based on two criteria:

 (1) They are characterised by often repeated transactions with similar features and pooling of risks; (2) Guarantors are able to estimate the average loss based on available statistics by using a probability-weighted concept. One-off guarantees are, on the contrary, individual, and guarantors are not able to make a reliable estimate of the risk of calls.
- 17.105Only standardised guarantee scheme requires special explanation in the System.

1. Standardised loan guarantee schemes

- 17.106Loan guarantees may be provided by a financial institution, including but not confined to insurance corporations. They may also be provided by government units. It is possible but unlikely that non-financial corporations may provide these sorts of guarantees; it is most unlikely that they would be provided by any unit to a non-resident unit. As indicated above, standardised guarantee schemes have much in common with non-life insurance. In the general case, similar recording is recommended as described below.
- 17.107 When a unit offers a guarantee, it accepts a fee and incurs a liability to meet the call on the guarantee. The value of the liability in the accounts of the guarantor is equal to the present value of the expected calls under the guarantee, net of any recoveries the guarantor receives from the defaulting borrowers. The liability is entitled provisions for calls under standardised guarantees.
- 17.108A guarantee may cover a multi-year period. A fee may be payable annually or up-front. In principle the fee should represent charges earned in each year the guarantee holds with the liability decreasing as the period gets shorter. In principle, the same sort of recording should be followed here as for annuities. The fee paid is earned as the future liability decreases. In practice, some units operating loan guarantees may have data only on a cash basis. This is inaccurate for an individual guarantee but the nature of the standardised guarantee scheme is that there are many guarantees of the same type,

- though not all for exactly the same time period nor all starting and finishing on the same dates. Unless there is reason to suppose that there is a major change in the nature of the guarantee holders over time, using cash based data should not introduce significant error.
- 17.109 Altogether six sets of transactions need to be recorded in respect of standardised loan guarantee schemes; two relating to the measurement of the production and consumption of the guarantee service, three relating to redistribution and one in the financial account. The value of the output of the activity, the property income to be attributed to the guarantee holder (whether creditor or debtor) and the value of the service charge are calculated in the manner described above for non-life insurance with the concepts of fees replacing premiums and calls under a standardised guarantee scheme replacing claims.
- 17.110The production and consumption transactions are as follows:
 - (a) The output is recorded in the production account of the sector or sub-sector to which the guarantor belongs.
 - The service may be paid for by either the borrower or the lender of the loan being guaranteed. When non-financial financial corporations, corporations, general government or non-profit institutions pay fees to obtain a guarantee, constitute intermediate the fees consumption, recorded in their production account. Any fees for guarantees payable households are part of final consumption expenditure, recorded in the use of income account.
- 17.111The redistributive transactions cover property income attributed to policyholders in respect of standardised guarantee schemes, net fees, and calls under standardised guarantee schemes.
 - (a) Property income attributed to policyholders in respect of standardised guarantee schemes is recorded as payable by the guarantor. It is recorded as receivable by the unit paying the fee. Both payable and receivables are recorded in the allocation of primary income account.

- (b) Net fees are calculated as fees receivable plus premium supplements (equal to the property income attributed to the unit paying the fee for the guarantee) less the value of the services consumed. These net fees are payable by all sectors of the economy and receivable by the sector of the guarantor.
- (c) Calls under standardised guarantee schemes are payable by the guarantor and receivable by the lender of the loan under guarantee, regardless of whether the fee was paid by the lender or the borrower. Both net premiums and claims are recorded in the secondary distribution of income account.

2. Loan guarantees provided by government

17.112Governments often offer loan guarantees for specific policy purposes. Export credit guarantees are one example. The guarantees may be issued by a unit within government that can be treated as a separate institutional unit. When this is so, the normal rules for the allocation of government units to either publicly controlled corporations or as part of government apply. If a guarantee unit charges fees that are

economically significant (in this case this may be equivalent to saying that most of the calls plus the administrative costs are covered by the fees charged), then this is a market activity of government. It should be treated as a financial corporation and transactions should be recorded as described above. If the fees cover most but not all the costs, the recording is still as above. The loss made by the agency offering the guarantees made be covered by government on a regular or intermittent basis but this is not passed on to those seeking the guarantees as a subsidy. Regular payments are recorded as a subsidy to the agency and intermittent payments, covering cumulated losses, are recorded as capital transfers only when such payments are made.

17.113In general, when a government unit provides guarantees without fees or at such low rates that the fees are significantly less than the calls and administrative costs, the unit should be treated as a non-market producer within general government. However, if government recognises the probability of having to finance some of the calls under the guarantees to the extent of including a provision in its accounts, a transfer of this size from government to the unit concerned and a liability of this amount (under provisions for calls under standardised guarantees) should be recorded.

Part 3 The treatment of pensions in the System

G. Types of pension schemes

- 17.114Pensions are provided to individuals in an economy under one of three mechanisms, via social security, via employment related schemes other than social security or via social Together, social security and assistance. employment related pension schemes other than social security constitute social insurance schemes. Although the benefits provided under social assistance and some social insurance schemes may be very similar, the key distinction is that social insurance benefits are only paid if the beneficiary participates in the social insurance scheme, participation being normally evidenced by the beneficiary or another on his behalf making qualifying contributions have been made. Social assistance is paid without qualifying contributions having been made though meanstesting may be applied to applicants.
- 17.115The means by which pensions are provided to persons in retirement varies considerable from one country to another. This part of chapter 17 describes the most common forms of pension provision made under social insurance schemes though not all aspects may apply to all countries. Pension provision made under social assistance is not discussed in this chapter but in chapters 8 and 9.
- 17.116Social insurance pensions in all countries are provided, if at all, in part by general government and in part by employers. The part provided by general government is called social security and the part by employers is called employment related schemes other than social security. The division between which pensions are provided by social security and which by other employment related schemes varies considerable from country to country with the consequence that the coverage and therefore national perceptions of what the term "social security" designates also vary considerably. In order to make clear the recommendations in the System, it is necessary to consider the types of coverage provided in different countries.

- 17.117The narrowest form of social security pension is very basic. The level may be fixed independently of the size of contributions (though not of the fact that contributions have been made for a given period of time). An employee's right to a pension under social security is often portable from one employee to another but for many people in low paid jobs, working temporarily or intermittently, it may be the only form of pension provision which they can expect to receive.
- 17.118By contrast, in some countries most or all pension provision may be made via social security. In this case government acts as an intermediary relative to the employer so that once the employer has handed over the contributions to the scheme paid by himself and the employees, government then takes on the risk of making the eventual payment. Government relieves the employer of the risk that the cost of pensions may be too great for his enterprise to meet and assures the population that pensions will be paid, though it may do so with the qualification that it may alter the pensions payable, amount of even retrospectively, if economic conditions so dictate.
- 17.119Pension schemes run by private employers are usually not subject to retrospective adjustments of the amounts payable, but there is a risk that the employer may be unable to pay because he has gone out of business. Increasingly, though, protections for the pension entitlements of individuals is becoming more common. Equally, the pension built up with one employer may not be transferable ("portable") to a new employer though this too is undergoing change. While social security may be, and very often is, financed on a pay as you go basis, without building up reserves for future liabilities, other employer schemes are increasingly likely to have reserves set aside and even if not accounting conventions may require them to recognise pension entitlements of present and past employees in their accounts.

17.120 Employment-related pensions, other than the most basic form of social security, are seen as part of the compensation package and negotiations between employees and employers may focus on pension entitlements as much as on current conditions of service and pay scales. Often pensions are provided by private employers from funds that the employers control or contract to a third party such as an insurance corporation. These funds may also

provide social benefits other than pensions, for example private medical coverage. In certain jurisdictions it is possible for a specialised unit to agree to assume responsibility for providing pensions for a number of employers in return for assuming the risk of ensuring adequate funding is available to make the promised pensions. Such an arrangement is called a multi-employer pension scheme.

H. Social security pensions

- 17.121It is common but not essential for both employers and employees to make contributions towards a social security pension. It is also common for the contributions to be compulsory. Social security pensions are frequently funded on a pay-as-you-go basis. That is the contributions receivable in a period are used to fund the benefits payable in the same period. There is no saving element involved, either for the government operating the scheme or for the individuals participating in it. No liabilities for the scheme are recognised in the main accounts of the System although concern is often expressed that benefits may exceed contributions and this situation is likely to worsen in an ageing population. reason, estimates of the liabilities of social security as well as any other pension schemes not included in the main accounts are included in a supplementary table described below in section J.
- 17.122The recording of the flows for social security pension schemes are simple. Any contribution
- made by the employer is treated as part of compensation of employees. It is recorded as payable by the employer in the generation of income account and receivable by the employee in the distribution of primary income account. The employee then pays an amount equal to what he receives from the employer together with any contribution he is liable to make on his own behalf to the social security fund. This amount is recorded as payable by households in the secondary distribution of income account and receivable by the government in the same Any contributions made by selfemployed or non-employed people are also included with the contributions payable by households to government. Social security benefits are also recorded as payable by government and receivable by households in the secondary distribution of income account.
- 17.123An example of these flows is shown in table 17.6. It is similar in content to table 17.1 except that table 17.1 relates to non-pension benefits and table 17.6 to pension benefits

I. Employment related pension schemes other than social security

- 17.124There are two forms of employment related pension schemes other than social security. One is called a defined contribution scheme, sometimes referred to as a money purchase scheme. The other is a defined benefit scheme, sometimes referred to as a final salary scheme, though this term does not accurately describe all defined benefit schemes. Typically both schemes are contributory, often by both the employer and the employee.
- 17.125A defined contribution schemes is one where the benefits are defined exclusively in terms of the level of the fund built up from the contributions made over the employee's working life and the increases in value that result from the investment of these funds by the manager of the pension scheme. The entire risk of the scheme to provide an adequate income in retirement is thus borne by the employee.

- 17.126A defined benefit schemes is one where the benefits payable to the employee on retirement are determined by the use of a formula, either alone or as a minimum amount payable. In this case the risk of the scheme to provide an adequate income in retirement is borne either by the employer or is shared between the employer and employee. In certain cases, the employer's risk may be borne by the multi-employer scheme that operates the defined benefit pension scheme on behalf of the employer. A scheme that may be defined in terms similar to a defined contribution scheme but with a guaranteed minimum, say, or other such hybrid schemes are grouped with defined benefit pension schemes in the System.
- 17.127 For both types of schemes, pension entitlements of the participants are recorded as they build up. In both cases, there is investment income earned on existing entitlements and this is recorded as being distributed to the beneficiaries and reinvested by them in the pension scheme. There are, though, a number of different features of the two schemes, so the transactions relating to each are described in detail separately before turning to other changes in the levels of pension entitlements. The recording of transactions for a defined contribution scheme is less complicated than the defined benefit scheme and is described first.
- 17.128For both types of schemes, a pension fund is assumed to exist. For a defined contribution pension scheme, a fund must exist. For a defined benefit pension scheme a fund may exist in reality or it may be a notional fund. If it exists, it may be part of the same institutional unit as the employer, it may be a separate institutional unit (an autonomous pension scheme) or it may be part of another financial institution, either an insurance corporation or a multi-employer pension scheme. In describing the recording of transactions, transactions with the pension fund must be attributed to the sector where the fund is located.

1. Defined contribution pension schemes

17.129 Recording the transactions related to a defined contribution pension scheme presents no conceptual problems. There are no associated imputations either for the flows concerned or for the values appearing in balance sheets for the pension entitlements of the beneficiaries nor

any doubt as to which unit has a liability and which an asset.

Transactions recorded for a defined contribution pension scheme

- 17.130The contribution made by an employer to a defined contribution pension scheme on behalf of his employee is treated as part of compensation of employees. It is recorded as payable by the employer in the generation of income account and receivable by the employee in the distribution of primary income account.
- 17.131The investment income on the cumulated pension entitlements is also recorded as being distributed to (receivable by) the employee in the distribution of primary income account and is shown as payable by the pension fund. The investment income includes interest and dividends payable plus the distributed income of collective investment schemes if the pension fund holds shares in them. It is possible that the pension fund may own property and generate net operating surplus on this which is also included along with the investment income as being distributed to the pension beneficiaries. In this case, the term investment income is to be interpreted as being elastic enough to include this source of income if it exists. Holding gains and losses generated by the investment of the cumulated pension entitlements are not included in investment income.
- 17.132 Part of the income distributed to the pension beneficiaries is used to meet the costs of operating the pension fund. This cost is shown as the output of the pension fund in the production account and as an element of consumption expenditure by households in the use of income account. The remaining part of the distributed income is treated as pension contribution supplements paid back by households to the pension funds
- 17.133 In the secondary distribution of income account, social contributions are shown as payable by households and receivable by the pension fund. The total amount of the social contributions payable is made up of the actual contributions payable by the employers as part of compensation of employees, actual contributions by employees and possibly by other individuals plus the contribution supplements just specified. Those other than

- employees who contribute to a defined contribution pension scheme may be self-employed persons participating in a defined contribution pension scheme or may be persons not employed who participate in a defined contribution pension scheme by virtue of their profession or former employment status, for example.
- 17.134Also in the secondary distribution of income account, the pension benefits payable to households by the pension fund are shown. However, the benefits payable under a defined contribution pension scheme take the form of a lump sum payable at the moment of retirement. It may be a requirement of the scheme that these sums are to be immediately converted to an annuity with the same or another financial institution but this is not a universal requirement. The appropriate recording of the benefits is to show the benefit as payable immediately on retirement and then, where appropriate, reinvested in terms of an annuity or other forms of financial assets.
- 17.135 For an individual, it is clear that the payment in full of pension entitlements is more in the nature of a withdrawal of saving than a current However, for a whole set of transfer beneficiaries, the number reaching retirement in a single year and withdrawing all their entitlements can be seen as an approximation to all retirees withdrawing a year's worth of entitlements. Because of the collective nature of social insurance, therefore, the amount of benefits payable in a year are recorded as current transfers. The consequential financial transactions, such as the purchase of an annuity, are recorded as an event not connected with pension provision. The recording of annuities is discussed elsewhere in this chapter.
- 17.136In the use of income account, there is an entry for the payment of the service provided by the

- pension fund (equal to the value of the pension fund's output) payable by households to the pension fund.
- 17.137In the same account there is an entry showing decrease) in pension increase (or entitlements caused by the excess of contributions payable less benefits receivable in the secondary distribution of income account. This amount is shown as payable to households by the pension fund. The rationale for this is that since this increase (or decrease) in pension entitlements directly affects the net worth of households, it should be included in the saving of the household sector. Because much of the increase in the pension entitlement of participants in a defined contribution pension scheme, and thus ultimately the funding for the benefits, come from holding gains that are not included in the contribution supplements of participants in defined contribution pension schemes, the adjustment for the change in pension entitlements for these individuals will frequently be negative.
- 17.138The adjustment for the change in pension entitlements that is included in the use of income account as payable by the pension fund to households is shown in the financial account as payable by households to the pension fund. The other factors affecting the change in the balance sheet entry for the change in pension entitlements are shown in the other changes in assets accounts and are discussed below in section XX.
- 17.139Table 17.7 illustrates the entries necessary to record the transaction related to a defined contribution scheme. It is simpler than the corresponding table for a defined benefit scheme which is described in the following section because of the absence of any imputed transactions.

Table 17.6: Accounts for pension benefits paid through social security - uses

Ш	lc	Δ

	Social security			Total
	Employer	fund	Households Other sectors	economy
Generation of income account				
Employers' actual social security contributions (pension)	139.0			139.0
Distribution of primary income				
Employers' actual social security contributions (pensions)				
Secondary distribution of income account				
Social security contributions (pensions)			226.0	226.0
Employers' actual social security contributions (pensions)			139.0	139.0
Household actual social security contributions (pensions)			87.0	87.0
Social security pension benefits		210.0		210.0

Table 17.7: Accounts for pension payable under a defined contribution scheme - uses

Uses

	Employer	Pension fund	Households	Other sectors	Total economy
Production account	Litipioyei	i chaon iunu	Householus	Other sectors	economy
Output					
Generation of income account					
Employers' actual pension contributions	14.0				14.0
Distribution of primary income	14.0				14.0
Employers' actual pension contributions					
Property income				3.0	
Property income payable on pension entitlements		3.0		3.0	
Secondary distribution of income account		5.0			
Household total pension contributions			27.1		27.1
Employers' actual pension contributions			14.0		14.0
Household actual pension contributions			11.5		11.5
Household pension contribution supplements			3.0		3.0
Pension scheme service charges			-1.4		-2.0
Defined contribution pension benefits		26.0	-1.4		26.0
Use of income		20.0			20.0
Final consumption expenditure			1.4		1.4
Change in pension entitlements		1.1	1.7		-0.5
Saving	-14.0	1.4	15.6	-3.0	0.0
Saving	14.0	1.4	10.0	3.0	0.0
Changes in assets					
Financial account					
Net borrowing/lending					
Change in pension entitlements			1.1		
Other financial assets	-14.0	2.5	14.5		
Balance sheet (start of year)					
Pension entitlements			140.0		
Pension fund assets		140.0			
Changes in the year					
Increases in pension entitlements (contributions)			27.1		
Decrease in pension entitlements (benefits)			-26.0		
Increase in assets		2.5			
Less operating surplus = own funds		-1.4			
Holding gains on pension fund assets		13.0			
Transfer of holding gains and losses to households			13.0		
Balance sheet (end of year)					
Pension entitlements			154.1		
Pension fund assets		154.1			

Table 17.6: Accounts for pension benefits paid through social security -resources

				Resources
	Social security			Total
	Employer fund	Households	Other sectors	economy
Generation of income account				
Employers' actual social security contributions (pension)				
Distribution of primary income				
Employers' actual social security contributions (pensions)		139.0		139.0
Secondary distribution of income account				
Social security contributions (pensions)	226.0			226.0
Employers' actual social security contributions (pensions)	139.0			139.0
Household actual social security contributions (pensions)	87.0			87.0
Social security pension benefits		210.0		210.0

Table 17.7: Accounts for pension payable under a defined contribution scheme - resources

Table 17.7. Accounts to pension pay					esources
	Employer	Pension fund	Households	Other sectors	Total economy
Production account					,
Output		1.4			1.4
Generation of income account					
Employers' actual pension contributions					
Distribution of primary income					
Employers' actual pension contributions			14.0		14.0
Property income		3.0			3.0
Property income payable on pension entitlements			3.0		3.0
Secondary distribution of income account					
Household total pension contributions		27.1			27.1
Employers' actual pension contributions		14.0			14.0
Household actual pension contributions		11.5			11.5
Household pension contribution supplements		3.0			3.0
Pension scheme service charges		-1.4			-1.4
Pension benefits			26.0		26.0
Use of income					
Final consumption expenditure					
Change in pension entitlements			1.1		-0.5
Saving					
				Chango	s in liabilities
Financial account				Change	3 III IIdbiiitic3
Net borrowing/lending	-14.0	1.4	15.6		
Change in pension entitlements	-14.0	1.1	13.0		
Other financial assets		1.1			
Balance sheet (start of year)					
Pension entitlements		140.0			
Pension fund assets				140.0	
Changes in the year					
Increases in pension entitlements (contributions)		27.1			
Decrease in pension entitlements (benefits)		-26.0			
Increase in assets	2.5			2.5	
Less operating surplus = own funds				-1.4	
Holding gains on pension fund assets				13.0	
Transfer of holding gains and losses to households		13.0			
Balance sheet (end of year)					
Pension entitlements		154.1			
Pension fund assets				154.1	

2. Defined benefit pension schemes

Differences between a defined benefit and a defined contribution pension scheme

- 17.140The fundamental difference in accounting for a defined benefit pension scheme as compared with a defined contribution pension scheme is that for the defined benefit pension scheme, the benefit to the employee in the current period is determined in terms of the undertakings made by the employer about the level of pension ultimately receivable, whereas for the defined contribution pension scheme the benefit to the employee in the current period is determined entirely by the contributions made to the scheme and the investment income earned on these and previous contributions. Thus while there is (in principle) exact information available on the benefits for the participant in the defined contribution pension scheme, the benefits for the participants in a defined benefit pension scheme must be estimated. The source of these estimates is the actuarial estimates the employer is faced with in drawing up his own accounts.
- 17.141 There are four sources of changes in pension entitlements in a defined benefit pension scheme. The first of these, the current service increase, is the increase in entitlement associated with the wages and salaries earned in the current period. The second source, the past service increase, is the increase in the value of the entitlement due to the fact that for all participants in the scheme, retirement (and death) are one year nearer. The third change in the level of entitlement is a decrease due to the payment of benefits to retirees of the scheme. The fourth source of change comes from other factors, factors that are reflected in the other changes in assets account.
- 17.142 As with a defined contribution pension scheme, both employer and employee may make actual contributions to the scheme in the current period. However, these payments may not be sufficient to meet the increase in the benefits accruing from the current year's employment. Therefore an additional contribution from the employer is imputed to bring equality between the contributions and the increase in current service entitlements. These imputed contributions are usually positive but it is possible for them to be negative if the sum of

- the contributions received exceed the increase in current service entitlements. The implications of this case are discussed below when examining the relationship between the employer and the fund.
- 17.143 At the end of an accounting period, the level of the pension entitlements due to past and present employees can be calculated by estimating the present value of the amounts due to be paid in retirement using actuarial estimates of the expected life length of the beneficiaries. This is the amount that appears in the balance sheet as the liability towards the employees. element in the increase of this amount year by year is the fact that the present value of the entitlements existing at the beginning of the year and still due at the end of the year have increased because the future is one year nearer and so one fewer discount factor must be used to calculate the present value. It is this unwinding of the discount that accounts for the past service increase in entitlements.
- 17.144A further basic difference between a defined benefit pension scheme and a defined contribution pension scheme concerns the payment for the cost of operating the pension scheme. As already noted, under a defined contribution pension scheme all the risk is borne by the beneficiaries. The pension schemes is operated on their behalf and they pay for the cost of it. Since the fund may be operated by a unit other than the employer, it is appropriate to treat the operating cost as part of the investment income that is retained by the fund to meet its costs (and generate a profit). In keeping with accounting for insurance, the investment income is treated as being attributed in full to the beneficiaries, part being used to meet the cost and the remained being reinvested with the fund.
- 17.145 For a defined benefit pension scheme, the situation is somewhat different. The risk that the fund may be insufficient to meet the promises of entitlement is met in part or in whole by the employer (or a unit acting on his behalf) and not by the beneficiaries alone. The fund may be directly controlled by the employer and be part of the same institutional unit or may be purely notional. Even in this case, there are costs associated with operating the scheme. Although there are initially borne by the employer, it is appropriate to regard this as a

form of income in kind provided to the employees and for convenience it may be included with the employers' contributions. (There is an element of pragmatism in this since this assumes all the costs are borne by current employees and none by retirees. It also assumes that the attribution that must be made in the case of notional schemes can be applied in other circumstances also.)

Transactions recorded for a defined benefit pension scheme

- 17.146The initial discussion assumes that the employer retains the whole responsibility for meeting the pension payments. Alternatives involving the use of a multi-employer scheme or where government assume responsibility on behalf of the employer are discussed subsequently.
- 17.147The total contribution made by an employer to a defined benefit pension scheme on behalf of his employee must be sufficient that, together with any actual contribution by the employee and excluding the cost of operating the scheme, it exactly matches the current service increase in the employee's pension entitlements. The contribution by the employer is divided into an actual and an imputed part, the latter being calculated so as to meet the need of an exact match between all contributions to the fund adding to the entitlements of the employee and the current service cost of these entitlements.
- 17.148The contribution by the employer should be calculated in relation to the pension entitlement earned in the period regardless of any investment income earned by the scheme in the same period or any over-funding of the scheme. The current period entitlement is part of compensation of employees and not to include the full value of the employer's contribution understates compensation of employees and therefore overstates operating surplus extreme case has occurred in the past when the investment of the pension entitlements has done so well that the employer has taken a "contribution holiday", that is he has not made actual contribution towards entitlements. It is important that contributions continue to be recorded even in the event of a contributions holiday, the benefit to the employer being regarded as a change in liabilities between the pension fund and the employer. This leaves the net worth of both the

- same as when contributions are not recorded under a contributions holiday without reducing compensation of employees artificially.
- 17.149 Under many defined benefit schemes, there is a qualifying period before an employee does in fact become eligible to receive a pension in retirement. Despite this qualifying period, contributions should be recorded from the start of employment adjusted by a factor reflecting the probability that the employee will in fact satisfy the qualifying period.
- 17.150The sum of employers' actual and imputed pension contributions are treated as part of compensation of employees. It is recorded as payable by the employer in the generation of income account and receivable by the employee in the distribution of primary income account.
- 17.151The increase in the present value of the entitlements of continuing employees (the past service increase) represents the investment income distributed to the employees. deduction is made for any amount that may be funded from holding gains or that is not actually matched by existing funds. It matches the amount that is unequivocally due to the employee under the prevailing agreements; the means by which the employer may ultimately match this obligation is not relevant for the recording of this as investment income any more than the means by which interest or dividend are actually financed affects their recording as investment income. The investment income is recorded as payable by the pension fund and receivable by households. It is immediately reinvested by the households in the fund and in this guise is described as pension contribution supplements.
- 17.152In the secondary distribution of income account, social contributions are shown as payable by households and receivable by the pension fund. The total amount of the social contributions payable is made up of the actual and imputed contributions payable by the employers as part of compensation of employees (excluding the amount of the costs of running the pension scheme), plus actual contributions by employees plus the contribution supplements just specified.

- 17.153 Also in the secondary distribution of income account, the pension benefits payable to households by the pension fund are shown.
- 17.154In the use of income account, there is an entry for the payment of the service provided by the pension fund (equal to the value of the pension fund's output) payable by households to the pension fund.
- 17.155 Also in the use of income account, there is an entry showing the increase (or decrease) in pension entitlements caused by the excess of contributions payable less benefits receivable in the secondary distribution of income account. This amount is shown as payable to households by the pension fund. The rationale for this is that since this increase (or decrease) is pension entitlements directly affects the net worth of households, it should be included in the saving of the household sector. In the case of a defined benefit pension scheme, the amount is unlikely to be negative unless it is scheme for a defunct employer and it is only paying benefits and not receiving new contributions.
- 17.156The same amount that is included in the use of income account as payable by the pension fund to households is shown in the financial account as payable by households to the pension fund. The other factors affecting the change in the balance sheet entry for the change in pension entitlements are shown in the other changes in assets accounts and are discussed below in section 4.

Defined benefit pension schemes operated by other than employers

17.157It is possible that some other organisation, such as a trades union, may operate a defined benefit pension scheme for its members that is in all respects parallel to an employer's defined benefit pension scheme. Exactly the same recording is followed as immediately described except that references to the employer should be understood to refer to the scheme organiser and references to the employee should be understood to refer to the participant in the scheme.

The relationship between the employer and the pension fund

- 17.158 As noted above, an employer may contract with another unit to manage the pension fund and arrange disbursements to the beneficiaries. There are two ways in which this may happen. The operator of the pension fund may simply act as the employer's agent and the responsibility for any short-fall in the fund (or the benefit of any excess) remains with the employer. However, it is not uncommon for a single unit to contract with several employers to manage their pension funds as a multi-employer pension fund. The arrangement are such that the multi-employer pension fund accepts the responsibility for any short-fall in the funds to meet the liabilities in return for the right to keep any excess funds. By pooling the risks over a number of employers the multi-employer fund expects to balance under and over funding so as to emerge with an excess over all the funds taken as whole in a similar way that an insurance corporation pools risk for many clients.
- 17.159In the case where the employer retains the liability for any under-funding or the benefit of any over-funding, a claim on (or liability towards) the employer by the pension fund should be recorded for any deficit or surplus. This is where the impact of a contribution holiday is recorded as a flow from the pension fund to the employer. The change in the claim or liability between the pension fund and the employer is recorded period by period as the difference between the investment income due to the beneficiaries and the investment income earned by the fund. Also any holding gains and losses on the assets managed by the fund are attributed to the employer so that the net worth of the pension fund remains exactly zero at all times.

3. A numerical example

- 17.160In order to illustrate the recording of transactions connected with a defined benefit pension scheme, Table 17.8 shows a numerical example. Figures that are imputed are shown in bold; those that result from re-routing are shown in italics.
- 17.161 Actuarial calculations show that the increase in pension entitlement coming from current

- service, that is the pension "earned" in the year in question is 15. Households (the employees) contribute 1.5. The employer therefore is obliged to provide 13.5. In addition the cost of operating the scheme is estimated at 0.6. In total therefore the employer must provide 14.1. He actually contributes 10 so the remaining 4.1 is an imputed contribution. The output of 0.6 is shown in the production account; the contributions by the employer are shown as payable by the employer in the generation of income account and receivable by the households in the distribution of primary income account.
- 17.162In the distribution of primary income accounts, property income is also shown. The increase in pension entitlement coming from past service, due to the unwinding of the discount factor because retirement is one year nearer, is 4. This is shown as an imputed flow of property income from the pension fund to households. At the same time, the pension fund actually earns 2.2 from investment income of the funds they manage. At this point, therefore, there is a shortfall of 1.8 in the pension fund resources but it is not shown in the current accounts.
- 17.163In the secondary distribution of income accounts, the payments from households to the pension fund are shown. These correspond to the amounts received in the distribution of primary income account except that the amount of service charge, the 0.6 of output, must be deducted. The total of the actual and imputed contributions is 13.5, the actuarial amount less the actual contribution by households.
- 17.164In the use of income account, as well as the purchase of the service charge as part of household consumption expenditure, the change in pension entitlement is shown as payable by the pension fund to households. accounts, no distinction is made between actual and imputed contributions but for illustrative purposes in this example the split is shown both for the adjustment item and the impact on saving. Here it can be seen that there is a shortfall of 7.5 in the pension entitlement. Of this 4.1 has been identified as the imputed actual contribution of the employer; the remaining 3.4 is equal to the property income attributed to households (4) less the output of the pension fund (0.6).

- 17.165 These figures are also shown in the financial account. (The capital account is not shown as there are no entries to be made there; the figures for saving carry through directly to net lending/borrowing.) The columns to note are those for the pension fund. The saving of 2.2 (equal to the property income earned from the rest of the economy) is augmented by 11.5, the "cash" payments of 10 by the employer and 1.5 by households, and diminished by the payment of pensions of 16, implying a decrease in other assets of 2.3.
- 17.166Turning to the balance sheet, the start of year account shows that total liabilities of the pension fund towards households was 170. The pension fund held actual assets of 130 and had a claim of the employer for the remaining 40. During the year, the increase in entitlements is 19 but pensions of 16 are actually paid so the end of year level of entitlements is 173. There are two causes of the increase in value of the assets at the disposal of the pension fund to meet these. One is the 2.3 decrease shown in the financial account, just discussed. The other is holding gains (to be shown in the revaluation account but not included in the flows table) of 12.0. Together, therefore, these exceed the 3 required by 6.7. Since the pension fund must always have a zero net worth, this 6.7 is shown as a reduction in the claim of the pension fund on the employer. How does it happen that although the employer did not pay sufficient contributions to cover the current service increase in pension entitlements and paid nothing towards the past service increase (4), far from increasing his debt to the pension fund he has reduced it? The answer is that because the pension fund must have zero net worth, although the employer is responsible for any shortfall in the fund's resources, he also benefits from any excess. In this case the initial shortfall of 7.5 (the difference between the increase in entitlements of 19 and the cash payments of 11.5) was more than compensated by the property income earnings of 2.2 and the holding gain of 12. Had the holding gain only be 3, say, then the claim of the pension fund on the employer would increase by 2.3 (7.5 - 2.2 - 3). It is this aspect of the importance of holding gains that makes offering to run a multiemployer scheme an attractive proposition.

Table 17.8: Accounts for pension payable under a defined benefit scheme - uses

	Employer	Pension fund	Households	Other sectors	Total economy
Production account					
Output					
Generation of income account					
Employers' actual pension contributions Employers' imputed pension contributions	10.0 4.1				10.0 4.1
Distribution of primary income					
Employers' actual pension contributions Employers' imputed pension contributions Property income Property income payable on pension entitlements		4.0		2.2	2.2 4.0
Secondary distribution of income account					
Household total pension contributions			19.0		19.0
Employers' actual pension contributions			10.0		10.0
Employers' imputed pension contributions			4.1		4.1
Household actual pension contributions			1.5		1.5
Household pension contribution supplements Pension scheme service charges			4.0 -0.6		4.0
Pension benefits		16.0	0.0		16.0
Use of income					
Final consumption expenditure Change in pension entitlements (actual) Change in pension entitlements (imputed) Saving (actual)	-10.0	-4.5 7.5 2.2	0.6 10.0	-2.2	0.6 -4.5 7.5 0.0
Saving (imputed)	-4.1	-3.4	7.5		0.0
Changes in assets					
Financial account					
Net borrowing/lending (actual)					
Net borrowing/lending (imputed) Change in pension entitlements (actual) Change in pension entitlements (imputed)			-4.5 7.5		
Pension fund claim on employer (current service)		4.1			
Pension fund claim on employer (past service)	10.0	3.4	115	0.0	0.0
Other financial assets (actual) Other financial assets (imputed)	-10.0 -3.4	-2.3 3.4	14.5	-2.2	0.0
·	-3.4	J. 1			
Balance sheet (start of year)			170.0		
Pension entitlements Pension fund assets		130.0	170.0		
Pension fund claim on employer		40.0			
Changes in the year					
Increases in pension entitlements (contributions)			19.0		
Decrease in pension entitlements (benefits)			-16.0		
Pension fund claim on employer		6.7			
Increase in assets		-2.3			
Holding gains on pension fund assets		12.0			
Balance sheet (end of year)			.=.		
Pension entitlements Pension fund assets		139.7	173.0		
Pension fund claim on employer		33.3			

Table 17.8: Accounts for pension payable under a defined benefit scheme - resources

				R	esources
	Employer	Pension fund	Households	Other sectors	Total economy
Production account					
Output		0.6			0.6
Generation of income account					
Employers' actual pension contributions Employers' imputed pension contributions					
Distribution of primary income					
Employers' actual pension contributions Employers' imputed pension contributions Property income Property income payable on pension entitlements		2.2	10.0 4.1 4.0		10.0 4.1 2.2 4.0
Secondary distribution of income account					
Household total pension contributions		19.0			19.0
Employers' actual pension contributions Employers' imputed pension contributions Household actual pension contributions		10.0 4.1 1.5			10.0 4.1 1.5
Household pension contribution supplements Pension scheme service charges Pension benefits		4.0 -0.6	16.0		4.0 -0.6 16.0
Use of income					
Final consumption expenditure Change in pension entitlements (actual) Change in pension entitlements (imputed) Saving (actual) Saving (imputed)			-4.5 7.5		-4.5 7.5
				Change	s in liabilities
Financial account					
Net borrowing/lending (actual)	-10.0	2.2	10.0	-2.2	0.0
Net borrowing/lending (imputed) Change in pension entitlements (actual) Change in pension entitlements (imputed)	-4.1	-3.4 -4.5 7. 5	7.5		0.0
Pension fund claim on employer (current service) Pension fund claim on employer (past service) Other financial assets (actual) Other financial assets (imputed)	4.1 3.4				
Balance sheet (start of year)					
Pension entitlements Pension fund assets Pension fund claim on employer	40.0	170.0		130.0	
Changes in the year					
Pension entitlements		19.0			
Decrease in pension entitlements (benefits) Pension fund claim on employer Increase in assets	-6.7	-16.0		-2.3	
Holding gains on pension fund assets				12.0	
Balance sheet (end of year)					
Pension entitlements Pension fund assets Pension fund claim on employer	33.3	173.0		139.7	

4. Other changes in the value of pension entitlements

17.167 For both defined contribution pension schemes and defined benefit pension schemes there are other changes to be recorded to the value of entitlements. Again the situation is much simpler for defined contribution pension schemes.

Defined contribution pension scheme

17.168The investment of the entitlements of defined contribution pension schemes lead to holding gains (and possibly losses). These come about through the management of the assets held by the fund but an amount exactly equal to the holding gains and losses should be attributed as an increase in the pension entitlement of the beneficiaries. This should appear as entries in the revaluation account.

Defined benefit pension scheme

- 17.169At first sight it would seem that there are no entries to be made in the other changes in assets accounts for a defined benefit pension scheme since the two components recorded as the pension contributions and investment income are matched exactly to the increase in entitlements. However, because the nature of a defined benefit pension scheme is that the amounts due are determined by a formula, there are other factors that may intervene to change the level of entitlements. These factors include a price escalation clause, changes in the formula used to determine benefits and demographic assumptions about life length. The special case of the impact of promotions on entitlements is discussed separately below.
- 17.170 A pension fund invests the funds at its disposal. If they work on a fully funded basis, the investment income should be more than enough to cover any price escalation clause in the pension agreement. The excess may also be sufficient to cover some other adjustments to entitlements. However, a major source of revenue comes from holding gains on investments. These were assumed to be sufficient to cover most or all changes in entitlements. It has become clear that many schemes were under-funded in the expectation that holding gains would also make up this shortfall also.

17.171 Given these adjustments are funded in large part by holding gains which appear in the revaluation account, it seems reasonable to record the contingencies that they are assumed to cover in the other changes in the volume of assets account except for the price escalation factor that should appear in the revaluation account.

The issue of promotions

- 17.172Many defined benefit pension scheme use a formula to set benefits that involves either the final salary or average salary as a key determinant. This implies that any promotion means that the total pension entitlements accrued to date are increased to take account of the new salary level. This is a significant benefit for the individual being promoted but what are the consequences for the employer's pension liabilities?
- 17.173The accounting profession uses two actuarial terms that bear on this discussion. The accrued benefit obligation (ABO) records, as it name implies, only the benefits actually accrued to date. It represents the amount the employee could walk away with if he left the firm tomorrow and may be the basis of assessing a person's net worth in the case of a divorce settlement, for example. A projected benefit obligation (PBO) is a more prudent measure of what the eventual level of entitlement is likely to be. For an individual, the PBO makes assumptions about how many future promotions the person is likely to receive and calculates his final salary accordingly. Then, if he has in fact only worked 20 out of an expected 40 years, it halves the final salary and calculates pension entitlement for the individual as if this were his current salary. Where an individual's ABO increases in steps as he is promoted, the PBO increases steadily over time. For the individual, PBO is always higher than ABO until the moment of retirement when the ABO catches up with the PBO.
- 17.174It would seem at first sight that the level of pension entitlements for a corporation should be the level of sum of all the pension entitlements of each of the employees and that therefore the sum of the PBO estimates would be considerably higher than that of the sum of the ABO and would evolve more smoothly over time. However, what is true for the individual

is not necessarily true for the cohort of employees. Suppose the employer has five classes of people for whose pensions he is responsible, four grades of employees and one set of retirees and for simplicity there are the same number of each. Consider the situation where in a year the retirees die; the most senior set of employees retire, the next three sets of employees are all promoted and a new set of employees is recruited at the lowest level. Every current employee is better off after promotion but the overall liability of the employer has not changed. This is analogous to the fact that every previous employee, still working for the employer, has been promoted but there are still the same number of people in the firm and still the same number at every grade. The effect of aggregating ABOs is to smooth the total entitlement and while it will still be lower than the aggregate PBOs, it will not necessarily be more volatile. Indeed it may be more stable.

17.175 While the profile of the ABO of an individual will show step changes when promotions occur, for a cohort of employees, the effect is much smoother. For a cohort of the same age remaining with the corporations for the whole of their working lives, the ABO estimates will be considerably lower that PBO estimates in the early years but the rate of increase of the ABOS will be faster than that of the PBOs so that at the point immediately before retirement, the two sets of estimates will be equal. Merging cohorts of employees with different periods of service with the corporation will bring the ABO estimates for all employees closer to the PBO one also.

17.176As long as the grade structure of the corporation stays the same, ABO and PBO will move roughly in step. If the firm expands and takes on many new employees at the lower grades, the PBO will be increase noticeably faster than the ABOs because the PBOs make estimates of how long the new employees will stay and how far they will be promoted while the ABOs record simply the pension accrued in their first year. If the firm decides to down size and reduces the number of their managerial staff, this will reduce the promotion prospects of the employees and a downward revision in PBO will be necessary. Because ABOs reflect simply the "locked-in" pension, this estimate is not affected.

17.177The question arises, though, of how to record the impact of promotion on the employee if an ABO recording is used. Any version of treating the increase as a form of compensation of employees or investment income falls back into the assumption that the aggregate of entitlements is the sum of the individual entitlements but without looking at other individual impacts on the aggregates such as when someone leaves and looses pension entitlement because not enough time has been served or when someone dies before retirement age. A simpler and adequate solution is to treat the rise in salary as a price change and record the change in the revaluation account.

17.178If the PBO method of recording entitlements is chosen as the preferred valuation, an adjustment in the other changes in volume of assets account is needed only if the structure of the enterprise changes so the chances of promotion change. On the other hand, the regular estimates of the employer's contributions to social insurance schemes included in compensation of employees will be systematically higher than those made under an ABO regime because the increase in pension entitlement that determines the size of the contributions will be based on a notional salary calculated on a PBO basis rather than the actual one.

5. Transferring pension entitlements

17.179One characteristic of the changing environment of pensions is the increasing possibility of having "portable pensions". Until recently it was often the case that a person leaving one employer had his pension frozen at that point and had to start a new pension with the new employer. It is becoming more common now for a person moving jobs to be able to convert the pension entitlement with the former employer to one with the new employer. When this happens, the pension entitlement of the household concerned is unaffected but there is a transaction between the two pension funds as the new one assumes the liability of the former. In addition there will be a counterpart transactions in some assets to match these liabilities. If the new employer is running a fund that is actually unfunded, he may receive cash from the former employer. If this cash is then used by the employer for purposes other than the pension fund, his liability to the fund

- increases and his use of the cash appears as net borrowing.
- 17.180If government assumes the responsibility for pension provision for the employees of a non government unit through an explicit transaction, a pension liability should be recorded in the balance sheet of government. If the government does not receive matching assets in return, the difference between the increase in the government's liability and the assets received is shown as a capital transfer to the non-government employer. There is further discussion of this type of arrangement in chapter 21.
- 17.181 Another way in which pension entitlements may be transferred between funds is when one corporation takes over another. In this case, assuming the take-over does not change the terms of the pension plan foe existing participants the transactions to be recorded for a group of employees (and retirees) is simply the aggregate of the position for each of the individuals.

6. A note on the tables

17.182 For cross-reference with tables in other chapters, table 17.9 shows the itemised components of transactions pertaining to social and other insurance in tables 17.1 to 17 .8 inclusive.

Table 17.9: Detailed transactions concerning social insurance

		Social			
Table number		Employer insurance fund	Households	Other sectors	Total economy
-	Property income payable on social insurance entitlements	11.0			11.0
17.5	Property income payable on non-pension entitlements	4.0			4.0
17.7	Property income payable on pension entitlements	3.0			3.0
17.8	Property income payable on pension entitlements	4.0			4.0
	Property income payable to insurance policy holders	13.0			13.0
17.3	Property income attributable to non-life insurance policy holders	6.0			6.0
17.4	Property income attributable to life insurance policy holders	7.0			7.0
	Secondary distribution of income account				
	Household total pension contributions		320.1		320.1
	Employers' social contributions		197.1		197.1
	Employers' actual social contributions		184.0		184.0
	Employers' actual social security contributions		154.0		154.0
17.1	Employers' actual social security contributions (non-pension)		15.0		15.0
17.6	Employers' actual social security contributions (pension)		139.0		139.0
17.5	Employers' actual non-pension contributions		6.0		6.0
17.7	Employers' actual pension contributions		14.0		14.0
17.8	Employers' actual pension contributions		10.0		10.0
	Employers' imputed social contributions		13.1		13.1
17.2	Employers' imputed non-pension contributions		9.0		9.0
17.8	Employers' imputed pension contributions		4.1		4.1
	Household actual social contributions		115.0		115.0
	Household actual social security contributions		97.0		97.0
17.1	Household actual social security contributions (non-pensions)		10.0		10.0
17.6	Household actual social security contributions (pensions)		87.0		87.0
17.5	Household actual non-pension contributions		5.0		5.0
17.7	Household actual pension contributions		11.5		11.5
17.8	Household actual pension contributions		1.5		1.5
47.5	Household social contributions supplements		11.0		11.0
17.5	Household non-pension contribution supplements		4.0		4.0
17.7	Household pension contribution supplements		3.0		3.0
17.8	Household pension contribution supplements		4.0		4.0
17.5	Social insurance scheme service charges Social insurance scheme service charges		-3.0 -1.0		-3.0 -1.0
17.5			-1.0 -1.4		-1.0 -1.4
17.7	Pension scheme service charges Non-pension service charges		-1.4 -0.6		-1.4 -0.6
17.0	Social benefits	296.0	-0.0	52.0	348.0
	Social security benefits	232.0		32.0	232.0
17.1	Social security non-pension benefits	22.0			22.0
17.6	Social security pension benefits	210.0			210.0
17.2	Unfunded non-pension benefits	9.0			9.0
17.5	Funded non-pension benefits	13.0			13.0
17.7	Defined contribution pension benefits	26.0			26.0
17.8	Defined benefit pension benefits	16.0			16.0
****	Social assistance	10.0		52.0	52.0

J. The special case of government providing pensions via social security

- 17.183In recognition of the fact that social security is normally financed on a pay as you go basis, entitlements accruing under social security (both pensions and other social benefits) are not normally shown in the System. If all countries had similar benefits provided under social security and under private schemes, international comparisons would be relatively straightforward. However, as pointed out at the beginning of this part, this is far from being the case and national perceptions of exactly what is covered by social security vary considerably.
- 17.184There are two problems with simply suggesting that entitlements from social security should be shown in the System. The first is that reliable estimates of the entitlements may not be readily available whereas it is increasingly the case that such estimates exist for private schemes. Secondly, there is an argument that such estimates are of limited usefulness where government has the possibility of changing the basis on which entitlements are determined in order to keep the entitlements within the bounds of what is budgetarily feasible. However, the consequence of simply accepting entitlements for private schemes are shown and for social security are not is that some countries would include the greater part of pension entitlements in the accounts and some would show almost none.
- 17.185In recognition of this dilemma, some flexibility regarding the recording of pension entitlements of unfunded pension schemes sponsored by government for all employees (whether private sector employees or government's own employees) is provided. Given the different institutional arrangements in countries, only some of these pension entitlements may be recorded within the main sequence of accounts (here referred to as the "core accounts"). In addition, however, a further table is to be presented that provides information disclosing the proportion of pension provision covered in the core accounts with some approximate estimates for the remaining schemes. It is a requirement, though, that a set of criteria be provided to explain the distinction between those schemes carried forward to the core accounts, possibly where the pension promise is of sufficient strength, and those recorded only in the supplementary table. By making this

- supplementary table and annotation a standard requirement for international reporting, analysts have the possibility of ensuring that cross-country comparisons are not unduly clouded by the institutional variations from country to country. Providing a single set of internationally recognized criteria for the distinction between the pension schemes fully recorded in the core accounts and those where the entitlements are shown only in the supplementary table is to be part of the SNA research agenda.
- 17.186The supplementary table is shown in table 17.??? As well as the possibility of including less robust estimates for countries with large social security sectors, the possibility will also exist of working back to a narrower coverage of private pensions for all countries being analysed.
- 17.187As noted above, providing detail on defined contribution schemes is relatively straightforward since full accounts must be available and no actuarial estimation is involved. Most of these are in the corporations sectors (column A) but it is possible that some government employees may be covered by them (column D). All defined contribution pension schemes should be included in the core accounts. Estimates for all defined benefit pension schemes outside social security should also be included (column B).
- 17.188 Government schemes for their own employees where separate accounting information, distinct from social security, is shown in the main accounts appear in columns E and F. Column E shows schemes managed by an insurance corporation and column F those managed by government itself. Any government schemes for their own employees distinct from social security which do not appear in the main accounts, are shown in column G. The sum of columns E, F and G therefore show the total responsibility of government for pension provision for their own employees. (Column F shows that part of all defined benefit schemes of government that are retained within the government accounts as distinct from being separated into separate units or managed for government by another institutional unit. Column H relates to social security schemes.

- Column C shows the total of all non-government schemes and column I the total of all schemes including social security.
- 17.189 For the most part, the beneficiaries of pension scheme are likely to be resident households. In some countries, though, the number of non-resident households receiving pension benefits may be significant. In this case, column J should be added indicating the amount of the total that concerns non-resident households.
- 17.190Some of the entries in the rows of columns G and H appear in the core accounts, even though the entitlements and change in entitlements do not, specifically the actual contributions made by both employers and employees. Other entries in the columns for G and H shown only in the supplementary table are shaded in the table and explained below.
- 17.191The imputed contribution by employers for those government schemes for which entitlements appear in column G but not in the core accounts requires special consideration. In

- the core accounts, this item is calculated, by convention, as equal to the difference between current benefits payable and actual contributions payable (by both employees and employers) In the supplementary table, this is replaced by the amount needed to ensure the total contributions, actual and imputed by employers and employees covers both the increase in pension entitlements from current service and the costs of operating the scheme.
- 17.192. An item calculated on the same basis in respect of social security is shown in Row 3 as "other (actuarial) accumulation of pension entitlements in social security funds". The distinction from employers' imputed social contributions is deliberate and is intended to emphasise the probable fragility of these estimates.
- 17.193 Items for household social contribution supplements, and the other changes in entitlements are shown on the same bases as for private schemes.

Table 17.10 A supplementary table showing the extent of pension schemes included and excluded form the SNA sequence of accounts

		Lia	bilities app	oear in the	core natio	onal accou	unts	appear in	es do not n the core accounts		
		Non-general government General government					nment				
		- 3	, , <u>J</u>				eral govern				
							ee define				
						0p.o)	schemes				
							3011011103				
											S
		Si			S	In the financial corporations sector	n the general government sector	In the general government sector	Social security pension schemes		Of which: Non-resident households
		эше			eme	SU	Jt Si	Jt Si	che		sno
		sch(nes		sche	atic	mei	mei	s uc	Se	甘
		on s	her		on s	Jod.	ern	erni	nsic	eme	ide
		outic	t sc		outic	COL	λog	ogo	be	sch	res
		ntri	nefi		Ē	cial	<u>ra</u>	<u>_</u>	ırity	non	lon
ē ē		00 F	l be		9	nan	ene	ene	seci	ensi	끍
qu		inec	inec	<u>-</u>	inec	ne fi	Je g	Je g	ial	al pe	vhic
Row number	Position / transaction / other flow	Defined contribution schemes	Defined benefit schemes	Total	Defined contribution schemes	<u>=</u>	<u></u> =	ln ‡	Soc	Total pension schemes	Of v
Roy	Column number	Α	В	С	D	Е	F	G	Н	I	J
	Opening balance sheet										
1	Pension entitlements										
	Transactions										
2	Social contributions relating to pension schemes										
2.1	Employer actual social contributions										
2.2	Employer imputed social contributions										
2.3	Household actual social contributions										
2.4	Household social contribution supplements										
	Other (actuarial) accumulation of pension entitlements in										
3	social security funds										
4	Pension benefits										
5	Change in pension entitlements										
	Change in pension entitlements due to transfers of										
6	entitlements										
	Other economic flows										
_	Changes in enttilemetns due to negotiated cahnges in										
7	scheme structure										
8	Revaluations Other shapes in values										
8	Other changes in volume Closing balance sheet										
9	Pension entitlements										
9	Related indicators										
-	Output										
	Assets held by pension schemes at end-year						1				
	Changes in pension entitlements related to revaluation of										
	assets held by pension schemes										
<u> </u>	ของบาง กับเน้ามา คุณกิจเบก จนกับก็ได้จ			L		<u> </u>	l				

Empty cells show where entries appear in the main ("core") accounts. Black cells show where no entry is appropriate. Grey cells show where information is provided in the supplementary table only.

Row 2 is the sum of rows 2.1 to 2.4

Row 3 is the analogue of employer's imputed contributions in the case where government has assumed the ultimate responsibility for any shortfall in pension provision

Row 5 is the sum of rows 2 and 3 less 4

More information on the components underlying rows 7 and 8 to be shown in a further supplementary table to allow an assessment of the degree of uncertainty in these estimates.

Part 4 The recording of flows associated with financial assets and liabilities

K. Introduction

- 17.194Financial corporations are distinguished from non-financial corporations because they play a particular role in the economy. Some facilitate means of payments between other units thus avoiding the need for barter. Some also provide the means whereby units seeking additional funds to finance capital formation, acquire financial assets or even for consumption can utilise the funding set aside by other units as saving. The equation that investment in capital formation must be equal to saving plus net borrowing from the rest of the world is fundamental to the functioning of the economy, the way financial markets work and so to the accounting system itself.
- 17.195Financial intermediation is the activity of matching the needs of borrowers with the desires of lenders. It is carried out by financial institutions preparing alternative sets of conditions under which clients can borrow and lend. These conditions allow for variations in the rate of return that may be expected from an investment with, often, higher returns being less certain than lower returns or involving forgoing access to the funds for a longer period. There are now very many, very diverse ways in which money can be borrowed and lent. The act of financial intermediation is thus one of devising financial instruments that encourage those with savings to commit to lend to the financial institutions on the conditions inherent in the instruments so that the financial institutions can then lend the same funds to others as another set of instruments with different conditions. This activity encompasses financial risk management and liquidity transformation.
- 17.196All financial intermediation in the System is carried out by financial institutions. However, some corporations in the financial sector are not themselves intermediaries but simply provide services auxiliary to financial intermediation.

For example, they may provide advice to clients about the terms available for specific types of borrowing and lending, such as a mortgage broker or provide certain sorts of financial resources such as a foreign exchange bureau that exchanges one currency for another.

1. Charging for financial services

- 17.197 Financial institutions provide services and charge for them. The means by which they charge, however, are not always obvious. When a bank offers "free banking" it only means that there are no explicit fees, not that there are no implicit fees. Fees may be charged indirectly by means of charging those purchasing a financial asset more than the seller of the same asset. For example, dealers in foreign exchange typically buy and sell at different rates; the differences between those prices and the mid-point represents a service charge paid by the customers.
- 17.198Nor is it only the service charge that may have to be measured indirectly. Bills are an offer of a fixed sum at some time in the future and the promise of this payment is sold at a discount. The increase in value between the buying price and the redemption price is treated as interest in the system.
- 17.199Nor are the terms in use in the financial markets exactly the terms used in the SNA. For example, the money paid by a bank on a deposit is described as interest by the bank but is not the amount recorded as interest in the SNA because the amount paid by the bank is assumed to be a compound payment representing interest as understood in the SNA less the service charges levied on the depositor for the costs of operating the account.

2. Explicit and implicit charges

17.200 Several financial institutions do make explicit fees for the services they render. These should always be recorded as payable by the unit to whom the services are rendered to the institution performing the service. services are rendered to a corporation or to government, the costs will form part of intermediate consumption. If they are rendered to households they will be treated as final consumption unless the financial service is performed in relation to an unincorporated enterprise, including the owning and occupying of a dwelling. Financial services are never incorporated into the value of any asset, financial or otherwise, even if their incurrence is necessary for the acquisition of the asset. (This is in contrast with the treatment on nonfinancial assets where the costs of acquiring the asset are included in the value of the asset appearing on the balance sheet.)

17.201 Implicit charges for services may be simply the difference between the buying and mid-price and between the mid-price and selling price as in the example of foreign exchange quoted above. (Each service should be calculated at the time of the transaction concerned so that holding gains and losses occurring between the time of the purchase and sale are not treated as services.) Other implicit charges may be combined with other transactions (or other flows) on a particular financial instrument. The service charge associated with borrowing and lending is one such example.

3. Investment income associated with financial instruments

17.202Most financial instruments give rise to investment income. Debt instruments such as

special drawing rights on the IMF (SDRs), loans, debt securities, deposits and some monetary gold where the amount is repaid according to a fixed formula give rise to interest. Equity and investment fund shares give rise to dividends or other distributions from corporate income. As far as possible, there should be no interest arising on other accounts receivable/payable since the outstanding that give rise to interest payments should be classified as loans. In practice this might not always be possible in which case there will be some amounts of interest shown under this instrument also. Except for other accounts receivable/payable, only gold bullion, financial derivatives and employee stock options never give rise to investment income.

4. Holding gains and losses on financial instruments

17.203Loans and deposits denominated in domestic currency do not give rise to nominal holding gains though there will always be real holding losses for the asset holder in the presence of inflation. Securities denominated in domestic currency where the income is in the form of coupons only may be subject to holding gains and losses. These occur because when the interest rate varies, the present value of the future coupon payments and terminal values change and this is reflected in the market price.

17.204For equity and investment fund shares, nominal holding gains are standard and may be substantial. Indeed, the most frequent reason for acquiring these instruments is in order to benefit from the holding gains that arise from holding them.

L. Recording flows in financial instruments

17.205 As explained above, both service charges and property income flows may be combined with the costs of acquiring and disposing of financial assets and liabilities. This section of the chapter, therefore, examines each class of instrument in turn to identify what flows should be recorded in each case. A summary of the types of flows that relate to each instrument is

given in table 17.X. Only the instruments relating to insurance, pension and standardised guarantee schemes are excluded as the treatment of these schemes is described in detail in other parts of this chapter.

1. Monetary gold

- 17.206Monetary gold consists of two sub-categories, gold bullion and unallocated gold accounts. Although it may not be possible to publish these two items separately for reasons of confidentiality, it is important to understand the different considerations that apply to each of them.
- 17.207 Gold bullion is physical gold owned by monetary authorities (or in certain cases other units authorised by them) as part of the reserve assets of a country. Gold bullion is not physically differentiable from commodity gold that may be held as a valuable by commercial banks or as inventories by some specialised industries, for example jewellers. Gold bullion may be sold by one monetary authority to another in another country. In such a case the exchange is recorded as an exchange of financial assets only. In all other cases, the gold is reclassified as commodity gold held by the monetary authority (and is no longer part of reserves) and is then sold as commodity gold. The reclassification is recorded in the other changes in the volume of assets account. If it is sold abroad it will feature in exports and imports of the countries concerned. commodity gold is sold, there may be a trade margin attached to it. When a monetary authority acquires monetary gold a reverse path is followed. The gold is acquired initially as commodity gold either from a domestic unit or from abroad and is subsequently reclassified to monetary gold.
- 17.208 There is no interest earned on gold bullion held as a commodity but it is subject to nominal and real holding gains and losses as the gold price alters. Interest can be payable when one monetary authority lends gold bullion held as reserves to another monetary authority.
- 17.209 Unallocated gold accounts are treated as foreign currency deposits unless they are held by the monetary authorities as part of foreign reserves. Unlike gold bullion, unallocated gold accounts have counterpart liabilities. Because the unallocated gold accounts classified as monetary gold must be held as part of foreign reserves, the counterpart liability is necessarily held abroad but it will be with a unit other than a monetary authority and thus the counterpart liability will not be treated as part of monetary

- gold in the counterpart country. (Assets held abroad as part of foreign reserves are generally not identified as such within the liabilities of the partner country.)
- 17.210Unallocated gold accounts attract interest and a service charge and are also subject to nominal and real holding gains and losses as the gold price alters.

2. SDRs

17.211SDRs are allocated to the countries and authorities participating in the SDR department of the IMF. Countries must be members of the IMF; other participants include a number of central banks, intergovernmental monetary institutions and development institutions. Participants may hold more or fewer SDRs than their allocation as a result of transactions in SDRs between participants. SDRs attract interest but no service charge as interest paid by participants holding more than their allocation exactly matches the interest owing to participants holding less than their allocation.. Data on the interest rates payable are available regularly from the IMF. Since the value of the SDR is based on a basket of four key currencies, the value of SDRs are always subject to nominal and real holding gains and losses. From time to time, new allocations of SDRs may be made; when this occurs the appearance is first recorded in the other changes in the volume of assets accounts and then the allocation is recorded as a transaction.

3. Currency

- 17.212Notes and coins are the simplest financial asset to record since for domestic currency, no service charges, property income or nominal holding gains and losses are recorded. Under inflation, though, the holder of notes and coins suffers real holding losses. Any charges made by a financial institution for operating a bank account, a fee for cashing a cheque or for withdrawing money from an automatic teller machine are all treated as explicit fees.
- 17.213Foreign currency should be recorded in the national balance sheets converted to a value in domestic currency using the exchange rate relevant for the date of the balance sheet. This value is subject to nominal and real holding gains and losses as the exchange rate of the

foreign currency relative to domestic currency alters. As noted above, there is usually a service charge associated with acquiring or disposing of foreign currency.

4. Deposits and loans

- 17.214In the 1993 SNA, financial intermediation other than that implicit in insurance activity was assumed to be restricted to deposits and loans and the only indirectly measured service charge was associated with interest flows on these instruments. The acronym used for the service flows became know as FISIM, Financial Intermediation Services Indirectly Measured. Although the update recognises other indirectly measured service charges associated with financial intermediation, it is convenient to continue to use the familiar expression, FISIM, for its traditional meaning, that is, for financial intermediation associated with loans and deposits.
- 17.215 Paragraphs 6.XX to 6.XX describe the basic principle of FISIM and explain the need to make the distinction, referred to above, between interest as understood by the banks holding deposits and issuing loans and the investment income flows recorded in the System. One (or possibly more) reference rate(s) should be applied to the level of loans and deposits to determine the SNA interest flows to be recorded. The difference between these flows and bank interest are recorded as service charges payable to the banks by the units holding the deposits or loans. This applies to both resident and non-resident units and to deposits and loans held with resident and nonresident units. For deposits with the banks, the service charge is equal to SNA interest less bank interest; for loans the service charge is equal to bank interest less SNA interest. At a minimum, it is probable that different reference rates should be used for every currency in which non-resident loans and deposits are denominated.
- 17.216No exclusion is made for lending of own funds. Although the act of lending, and the charging of SNA interest, is not a productive activity, there is a service charge associated with lending. A person borrowing from a bank is unaware of whether the amounts borrowed are of intermediated funds or come from the bank's own funds and no difference in the service

- charges applied should be made. Similarly, if a person borrows from a money lender, there is a service charge payable. (Often in fact service charge is very large reflecting the much higher risk of default faced by the money lender. A noteworthy feature of some micro-finance schemes is that, because defaults are uncommon, the charges are modest.)
- 17.217It is not always simple to determine whether holdings between banks should be classified as deposits or loans. In a complete flow of funds presentation, this should be resolved but in the absence of a flow of funds analysis, inter-bank positions may be shown under currency and deposits. By convention they are shown as deposits less loans with a negative sign if necessary. It is assumed that the inter-bank rate at which banks borrow and lend to one another is usually such as to meet the criteria for a reference rate, that is, it is a risk-free rate. (In some cases it may be appropriate to use the inter-bank rate as the reference rate.) For this reason, it may often be appropriate to assume that there is no FISIM associated with interbank lending and borrowing within the national economy.
- 17.218The outstanding balance on a credit card or on an account with a retailer is often subject to interest. These outstanding balances should be classified as loans, not other accounts receivable/payable, and FISIM should be calculated on them.
- 17.219 Repurchase agreements are classified as giving rise to deposits or loans depending on whether they are or are not included in national measures of broad money. They give rise to interest that may have a FISIM component but, even if this is so, it may be very difficult to isolate it. They do, though, have fees associated with their initiation and thus for pragmatic reasons the measurement of flows associated with them may be treated in the same way as for debt securities.
- 17.220 There are no nominal holding gains and losses on deposits and loans expressed in domestic currency (whether these are held by residents or non-residents). With any inflation at all, there will be real holding losses on assets denominated in domestic currency. There may be nominal and real holding gains and losses on deposits and loans denominated in other

currencies or held as unallocated metal accounts.

17.221The special case of non-performing loans and how they should be treated in the System is discussed in chapter 13.

5. Debt securities

17.222In terms of recording the associated flows, there are three types of debt securities. The first is where the amount payable at the end of the period for which the security exists is the same as the initial amount paid for the security but there are associated "coupons" that entitle the holder to payments of interest at intervals during the life of the instrument. The second type of security is one where no intermediate payments are made but the initial price is lower than the terminal price. The initial price is equal to the terminal price discounted to the date of issue at the appropriate rate of interest that could be earned on a deposit of similar characteristics. The increase in value of the security during its life is treated as interest accruing to the holder of the security that is "reinvested" in the security to increase its value. The third type of security is a hybrid of the two other forms; the initial value is less than the terminal value but there are also attached In certain circumstances, if the coupons represent a rate of interest higher than that currently on offer, the security may be offered at a price higher than the terminal price.

Service charges associated with securities

- 17.223 For securities, the interest calculated according to the coupon or as the increase in value of the security is recorded in the System as such without adjustment for a service charge. However, there is a service charge associated with the acquisition of a security on initiation and with the disposal and acquisition of a security at any point during its life. These service charges are identified as being the difference between the buying (bid) and selling (ask or offer) price quoted for each security and the mid-price.
- 17.224Suppose an instrument is bought for 102 and subsequently sold for 118 even though there has been no change in the rate of interest (and hence of the value of the instrument due to holding gains and losses). At first sight, it seem that

interest of 16 should be recorded. However, suppose the mid-price on purchase were 100 and on sale were 120. The correct recording would be to show interest of 20 with a purchase of services of 4. Ignoring the bid-ask spread understates interest and ignores the services provided by the financial intermediaries that buy and sell securities.

Valuing securities

- 17.225 There are two ways in which the value of a discounted security can be determined during its life when the prevailing interest rate is different from the rate prevailing when the security was initiated. The debtor approach is the perspective of the unit issuing the security and the creditor approach is the perspective of the unit holding the security. The first option, called the debtor approach, is to continue to use the rate prevailing on initiation throughout the instrument's life. The alternative, the creditor approach, is to use the current rate to estimate the value of interest between any two points in the instrument's life.
- 17.226Suppose an instrument if offered at 90 with a terminal value of 100. If the discount (interest) rate does not change during its life, interest will accrue steadily throughout. Suppose, though, that the interest rate falls when the instrument has reached a value of 95. Because the terminal value is now discounted by a smaller factor, the value of the security increases, say to 97. Both the creditor and debtor approach would record interest of 5 in the period before the interest rate fall. Under the creditor approach, this increase of 2 is treated as a holding gain and only the subsequent rise to the terminal value of 100 is treated as interest. Thus over the whole life of the instrument it has given rise to interest of 8 and a holding gain of 2.
- 17.227 In the System, the debtor approach is used. Under this approach, the interest accruing in the period before the interest rise is still 5 but so is the interest in the period after the interest rate rise. Adding this level of interest to the value of 97 when the rise occurred would give a value of 102 at the terminal date. Since this value is too high, a holding loss if 2 has to be noted. Thus over the whole life of the instrument there is interest of 10 with an initial holding gain of 2 (when the interest rate changed) offset by the later holding loss of 2. The rationale for using

the debtor approach is that the debtor, the issuer of the security, is not liable to make the payment until the security matures and from his perspective it is appropriate to treat the total amount of interest as accruing steadily over the life of the security.

Determining interest flows on bills and bonds

Interest on bills and similar instruments

- 17.228 Bills are short-term securities that give the holder (creditor) the unconditional right to receive a stated fixed sum on a specified date. They are issued and traded in organized markets at a discount that depends on current market short-term interest rates and the time to maturity. Most bills mature after a period ranging from one month to one year.
- 17.229 As the bill approaches maturity, its market value increases because there is less discounting applied to it. This increase in value, in common with the increase in the value of any asset due to the unwinding of a discount factor, is treated as income in the System. For financial assets, the income is recorded as interest.
- 17.230Let the price paid for a bill at its time of issue be L: this represents the amount of funds that the purchaser (creditor) provides to the issuer (debtor) and measures the value of the initial liability incurred by the issuer. Let the face value of the bill be F: this represents the sum paid to the holder of the bill (the creditor) when it matures. The difference, F-L, or discount on the bill, measures the interest payable over the life of the bill.
- 17.231 Bills are traded on money markets at values that gradually rise to reflect the interest accruing on the bills as they approach maturity. The increase in the value of a bill due to the accumulation of accrued interest does not constitute a holding gain because it is due to an increase in the principal outstanding and not to a change in the price of the asset.

Interest on bonds and debentures

17.232Bonds and debentures are long-term securities that give the holder the unconditional right to:

- (a) A fixed or contractually determined variable money income in the form of coupon payments; or
- (b) A stated fixed sum on a specified date or dates when the security is redeemed; or
- (c) Both (a) and (b). Most bonds fall into this category.
- 17.233The amounts of the fixed or variable money incomes or coupon payments due for payment within the accounting period are treated as interest receivable and payable by the creditor and debtor, respectively. In addition, when a bond is issued at a discount, the difference between the face value, or redemption price, and the issue price constitutes interest that accrues over the life of the bond, in the same way as for a bill. However, as accounts are compiled for time periods that are typically much shorter than the life of the bond, the interest must be distributed over those periods. The way in which this may be done is explained below.

Zero-coupon bonds

17.234Zero-coupon bonds are long-term securities that are similar to bills. They do not entitle their holders to any fixed or variable money income but only to receive a stated fixed sum as repayment of principal and accrued interest on a specified date or dates. When they are issued they are usually sold at a price that is substantially lower than the price at which they are redeemed on maturity. Let L equal the issue price and F the redemption price, so F-L is the value of the interest receivable and payable over the life of the bond. This interest has to be distributed over the years to its maturity. One possible method is to assume that interest at a rate of r is credited at the end of each year at an annual rate that is constant over the life of the bond, so that the final value $F = L(1+r)^n$

The interest rate, r, is given by the following expression:

$$r = \left(\frac{F}{L}\right)^{\frac{1}{n}} - 1$$

where n is the number of years from the time of issue to maturity. The interest accruing during the course of year t is then given by

 $rL(1+r)^{-1}$ where t = 1 at the end of the first year.

The interest accruing each year is effectively reinvested in the bond by its holder. Thus, counterpart entries equal to the value of the accrued interest must be recorded in the financial account as the acquisition of more bond by the holder (creditor) and as a further issue of more bond by the issuer (debtor).

Other bonds, including deep-discounted bonds

- 17.235 Most bonds pay a fixed or variable money income and may also be issued at a discount or, possibly, a premium. In such cases, the interest receivable by the holders of the bonds has two components:
 - (a) The amount of the money income receivable from coupon payments each period; plus
 - (b) The amount of interest accruing each period attributable to the difference between the redemption price and the issue price.

The second component is calculated in the same way as for zero-coupon bonds, as described above. In the case of deep-discounted bonds, the amounts of money income payable each period are relatively small and most of the interest accruing is attributable to the difference between the redemption price and the issue price. At the other extreme, some bonds offer an income stream in perpetuity and are never redeemed.

Index linked securities

17.236Index linked securities are financial instruments for which the amounts of the coupon payments (interest) and/or the principal outstanding are linked to a general price index, a specific price index, the price of a commodity or an exchange rate index. Different treatments are recommended for the recording of transactions depending on the type of index used to uprate the level of principal to which the interest is

linked and on the currency in which the interest and principal are denominated.

- 17.237 When the coupon payments are index linked and denominated in domestic currency, the full amounts of such payments are treated as interest receivable and payable, in the same way as the interest receivable and payable on any other security paying a contractually agreed variable income.
- 17.238When the value of the principal is linked to a broad-based index and expressed in domestic currency, in principle, the difference between the eventual redemption price and the issue price is treated as interest accruing over the life of the asset in the same way as for a security whose redemption price is fixed in advance. Because the final redemption price is not known before redemption, a strict application of this principle means that the interest payable each year can only be approximated until redemption. In practice, to avoid repeated revisions, the change in the value of the principal outstanding between the beginning and end of a particular accounting period due to the movement in the relevant index may be treated as interest accruing in that period, in addition to any interest due for payment in that period. The series is then not subject to revisions due to subsequent movements in the index.
- 17.239If the value of the principal is linked to an index that is volatile, for example the link is to a commodity price such as petroleum or gold, the standard procedure gives one of two problematic outcomes. Either significantly large revisions may have to be made throughout the life of the instrument or, if no revisions to prior years are accepted, then the path of interest payables can be very volatile and may include negative entries. To avoid these alternatives. modified approach At the end of each period, recommended. interest should be calculated on the basis of the value of the index at the redemption date expected at the end of the relevant accounting period. Again no retrospective revisions are to be made. A consequence is that the total amount of interest over the life of the instrument will not necessarily be equal to the difference between the issue price and redemption price. This treatment

- recommended as being preferable to the alternatives just described.
- 17.240 If the principal but not the coupon is indexed to a foreign currency, changes in the principal due to variations in the exchange rate are treated as holding gains and losses and not as interest. If both the principal and coupons are indexed to a foreign currency, the transactions relating to both the principal and coupons should be calculated by reference to the foreign currency even if the currency of settlement is different. In this case, variations in the level of the security due to currency fluctuations is treated as interest.
- 17.241 As with other securities, the interest accruing as a result of indexation is effectively reinvested in the security and these additions to the value of the security must be recorded in the financial accounts of the holder and issuer.

6. Equity and investment fund shares

- 17.242The financial service charges levied on transactions in equity and investment fund shares are calculated in the same away as for debt securities, that is, as explicit fees and the difference between the financial intermediary's selling price and the mid-price and the mid-price and the intermediary's buying price.
- 17.243The investment income from corporate equity takes the form of distributed income of corporations. For corporations, the distributed income is in the form of dividends. For quasicorporations, the investment income is withdrawals from income of quasi-corporations. As noted in chapter 7, dividends or other withdrawals from corporate income recorded as investment income at the time they are declared payable and not when earned. A different recording is made for extraordinarily large dividends that are out of line with recent experience on the amount of income available for distribution to the owners of the corporation. Any excess distribution is to be recorded as a withdrawal of equity (recorded in the financial account) and not as part of investment income. Chapter 21 discussed the case of exceptional dividends of public corporations.
- 17.244For investment funds, the income element comes in the form of investment income

- disbursements to collective investment fund share holders.
- 17.245For foreign direct investment enterprises, there will also be investment income in the form of reinvested earnings.
- 17.246As noted earlier, there may be considerable holding gains and losses, both nominal and real on equity and investment fund shares.
- 17.247 The entries in the financial accounts relating to acquisitions of equity contain two distinct types of transactions. One is the exchange of equity investment fund shares between institutional units. Because the transactions are valued at mid-price, total acquisitions must be equal to total disposals. The net effect, therefore, is to show the change in composition of the holders of shares by institutional sector and with the rest of the world. The second type of transactions included in the financial account is the receipt of reinvestment of earnings, the counterpart of the outflow recorded under investment income payable by corporations. In calculating the revaluation element between opening and closing balance sheet, care must be taken to exclude the reinvestment of earnings term.

7. Financial derivatives

17.248 Arranging a financial derivative may involve a set-up fee that should be shown as an explicit fee charged by the financial institution concerned and payable by the holder of the financial derivative. The initial value of a forward type financial derivative is zero but as soon as a change in circumstances, that the financial derivative is designed to provide protection against, occurs, the financial financial derivative will acquire a value. This change in value is shown in the revaluation account. At inception, options have a positive value normally equal to the premium paid to establish them. Thereafter, any change in value is recorded in the revaluation account. There is no investment income accruing on a financial derivative.

8. Employee stock options

17.249 As explained in chapter 7, the granting of a stock option may form a part of compensation of employees.

- 17.250The costs of administering stock options are borne by the employer and are treated as part of intermediate consumption just as other administrative functions associated with compensation to employees.
- 17.251Three dates are important in measuring the income from stock options. The grant date is when the employee is informed that a stock option will be available for him to exercise at some time in the future. The vesting date is the earliest date at which the option can be exercised and the exercise date is when the option is actually exercised.
- 17.252If possible, the value of the option at grant date is treated as part of the compensation of employees at that date and increases in the value between the grant date and vesting date are treated as further elements of compensation of employees. If it is not possible to spread the earnings between grant date and vesting date, the whole of the value may have to be recorded at vesting date. If there is a gap between vesting date and exercise date, and the value of the option changes in this period, it is recorded as a holding gain or loss of the employee.
- 17.253 Although the value of the stock option is treated as income, there is no investment income associated with the stock option.

17.254In certain circumstances a stock option can also be used as a form of payment for goods and services or as a contribution by the employer to the pension fund of his employees. In the former case, the stock options are included with employee stock options by convention even though they are not strictly related to employment.

9. Other accounts receivable/payable

- 17.255 Other accounts receivable/ payable are essentially accrual adjustments that show items that became liable for payment earlier but have not been paid. Trade credit refers to the case where goods and services have been delivered but payment has not yet been received. Advances refer to payment for work-in-progress for which pre-payment has been made but the goods are not yet delivered. The means of financing payment, such as the use of credit cards, is not included here; the balance on the cards is treated as a loan and payments such as interest or overdue fees are recorded as for loans.
- 17.256Other accounts payable/receivable denominated in domestic currency can have no nominal holding gains and losses but may have real ones. Any items denominated in foreign currency may have both nominal and real holding gains and losses.

Table 17.11 Indications of the flows associated with different financial instruments

	Property	Revaluation account					
Financial instrument	Buy/sell margin	Margin on interest	Interest	Dividends	Withdrawals from incomes of quasi- corporations	Investment income attributed to collective investment fund share holders	Nominal holding gains and losses
Monetary gold and SDRs	ina giii	I III CO COC	III.tor oot	Dividorido	joo.pordor.o	110.0010	41.4.100000
Monetary gold		X					x
Unallocated gold accounts			x				x
SDRs			x				x
Currency and deposits							
Currency							
Domestic							
Foreign	x						x
Transferable deposits							
Domestic		X	x				
Foreign		X	x				x
Interbank positions		(x)	x				
Other deposits							
Domestic		X	x				
Foreign		X	x				x
Debt securities	x		x				(x)
Loans							
Domestic		X	X				
Foreign		X	X				x
Equity and investment fund shares							
Equity							
Listed shares	x			X			x
Unlisted shares	x			X			x
Other equity					X		x
Investment fund shares/units							
Money market fund shares/units	x					X	x
Other investment fund shares/units	x					X	x
Financial derivatives and employee							
stock options							
Financial derivatives							x
Employee stock options							x
Other accounts receivable/payable			x				

Part 5 Contracts, leases and licences

M. Introduction

17.257 Many transactions that take place in the economy and are recorded in the System are specified in terms of a contract between two institutional units. The majority of contracts are such that one unit provides a good, service or asset to the other unit for an agreed payment at an agreed time (possibly immediately after agreeing on the price). Such contracts may be written and legally binding or may be informal or even only implicit. If a unit accepts the estimate provided by a builder for the cost of specified work, the contract is written and may well be legally binding. If a book is ordered from a bookshop but there is a delay in delivery, there is an informal contract between the book shop and the customer but it is unlikely to be enforceable by either side. Whenever a customer asks how much a given service will cost, whether it is a haircut, the delivery of a heavy product or entry to a cinema, accepting the service at the quoted price is in effect an implicit contract. However, all these contracts are simply agreements about the terms under which goods, services and assets are provided to the customer along with the legal ownership of The only extent to which these the item. contracts feature in the System is that the time of delivery agreed in the contract is the point at which the transaction is to be recorded in the accounts and not the time of payment. Any difference between the time of payment and time of delivery gives rise to an entry in the financial account under other accounts receivable/payable.

17.258However, there are other contracts and legal agreements variously described as leases and

licences (or permits) where the terms of the agreement may affect the time of recording of transactions made under the agreement as well as the classification of payments and the ownership of the item subject to the agreement. The purpose of this part of the chapter is to provide guidance on how transactions made under these more complex arrangements are to be recorded in the System.

17.259The first item for discussion concerns the different sorts of leases recognised in the System. The next topic for discussion is the treatment of permits to use natural resources. This is of particular importance when it is government that claims ownership of the resource on behalf of the community at large but can apply to privately owned resources also. This leads naturally into a discussion of the treatment of assets where more than one unit has a claim to ownership, or the benefits of ownership accrue to more than one unit.

17.260 Some contracts are not connected with the use of assets. The first for discussion are licences (or permits) given to undertake particular activities independently of any assets that may be used in the activity. Here there are different treatments when the permits are issued by government and when they are given by other institutional units. The next point for consideration is when a contract itself can constitute an asset in itself, independently of the subject of the contract. Finally, a number of clarifications are made concerning the timing and nature of payments made under a contract.

N. Leases

17.261Three types of leases are recognised in the System; operating leases, financial leases and

resource leases. Each of these leases relates to the use of a non-financial asset. Fundamental

to the distinction between the different sorts of leases is the difference between legal and economic ownership. This distinction is elaborated in chapter 3. The legal owner of an asset is the institutional unit entitled in law and sustainable under the law to claim the benefits associated with the asset. By contrast, the economic owner of an asset is entitled to claim the benefits associated with the use of the asset in the course of an economic activity by virtue of accepting the associated risks. The legal owner is often the economic owner also. When they are different, the legal owner has divested itself of the risks in return for agreed payments from the economic owner.

1. Operating leases

- 17.262An operating lease is one where the legal owner is also the economic owner and accepts the operating risks and receives the economic benefits from the asset by using it in a productive activity that requires him to provide any necessary repair and maintenance of the asset. Under an operating lease the asset remains on the balance sheet of the lessor.
- 17.263 The payments made under an operating lease are referred to as rentals and are recorded as payments for a service. The service provided by the lessor goes beyond the mere provision of a piece of equipment. It includes other elements such as convenience and security, which can be important from the user's point of view. The lessor, or owner of the equipment, normally maintains a stock of equipment in good working order that can be hired on demand or at short notice. The lessor must normally be a specialist in the operation of the equipment, a factor that may be important in the case of highly complicated equipment, such as computers, where the lessee and his employees may not have the necessary expertise or facilities to service the equipment properly themselves. The lessor may also undertake to replace the equipment in the event of a serious or prolonged breakdown.
- 17.264 Operating leasing developed originally to meet the needs of users who require certain types of equipment only intermittently. Many operating leases are still for short periods though the lessee may renew the rental when the period expires and the same user may hire the same piece of equipment on several occasions.

However, with the evolution of increasingly complicated types of machinery, especially in the electronics field, the servicing and back-up facilities provided by a lessor are important factors that may influence a user to rent. Other factors that may persuade users to rent over long periods rather than purchase are the consequences for the enterprise's balance sheet, cash flow or tax liability.

2. Financial leases

- 17.265 A financial lease is one where the lessor as legal owner of an asset passes the economic ownership to the lessee who then accepts the operating risks and receives the economic benefits from using the asset in a productive activity that requires him as lessee to provide any necessary repair and maintenance. Under a financial lease, the legal owner is shown as issuing a loan to the lessee with which the lessee acquires the asset. Thereafter the asset is shown on the balance sheet of the lessee and not the lessor; the corresponding loan is shown as an asset of the lessor and a liability of the lessee. Payments under the financial lease are treated not as rentals but as the payment of interest and repayment of principal.
- 17.266Very often the nature of the asset subject to a financial lease may be quite distinct from the assets used by the lessor in his productive activity, for example a commercial airliner legally owned by a bank but leased to an airline. It would make no economic sense to show either the aircraft or its consumption of fixed capital in the accounts of the bank or to omit them from the accounts of the airline. The device of a financial lease avoids this undesirable form of recording the ownership of the aircraft and the decline in its value while keeping the net worth of both parties correct throughout the length of the lease.
- 17.267 It is common for a financial lease to be for the whole of the life of the asset, but this need not necessarily be so. When the lease is for the whole of the life of the asset, the value of the imputed loan will correspond to the present value of the payments to be made under the lease agreement. This value will cover the cost of the asset and include a fee charged by the lessor. Payments made regularly to the lessor should be shown as a payment of interest, a payment for a service and a repayment of

capital. If the terms of the agreement do not specify how these three items are to be identified, the repayment of principal should correspond to the decline in the value of the asset (the consumption of fixed capital), the interest payment to the return to capital on the asset and the service charge to the difference between the total amount payable and these two elements.

17.268 When the lease is for less than the whole life of the asset, the value of the loan should still be estimated as the value of the asset plus the value of the service charges to be made under the terms of the lease. At the end of the lease, the asset will appear on the balance sheet of the lessee and its value will be equal to the value of the loan owed to the lessor at that time. At that

point the asset could be returned to the lessor to cancel the loan or a new arrangement, including the outright purchase of the asset, may be reached between the lessor and lessee.

3. Resource leases

17.269 A resource lease is one where the legal owner of a natural resource that has an infinite life makes it available to a lessee in return for a regular payment recorded as property income and described as rent. The resource continues to be recorded on the balance sheet of the lessor even though it is used by the lessee. Because the resource has an infinite life, there is no consumption of fixed capital recorded by either the lessor or lessee.

O. Licences and permits to use a natural resource

- 17.270As noted above, in many countries permits to use natural resources are generally issued by government since government claims ownership of the resources on behalf of the community at large. However, the same treatments apply if the resources are privately owned.
- 17.271 There are basically three different sets of conditions that may apply to the use of a natural resource. The owner may permit the resource to be used to extinction. The owner may allow the resource to be used for an extended period of time in such a way that in effect the user controls the use of the resource during this time with little if any intervention from the legal owner. The third option is that the owner can extend or withhold permission to continued use of the asset from one year to the next.
- 17.272 The first option results in the sale (or possibly an expropriation) of the asset. The second option leads to the creation of an asset for the user, distinct from the resource itself but where the value of the resource and the asset allowing use of it are linked. The third option comes back to the treatment of the use as a resource lease. The difference in treatment between the second and third options was articulated in the context of the case of a mobile phone licence and that recommendation (ref to NN) is recapitulated before seeing how each of the

three options relate to different types of natural resources.

1. The "mobile phone" treatment of licences or permissions to use a natural resource

- 17.273 The case arose in 2000 when the sale of licences to use radio spectra for third generation mobile phones brought a flurry of interest from companies wanting to have exclusive access to the spectra and who in consequence were prepared to bid (often by auction) extremely large sums for the access rights to the spectra.
- 17.274Eight conclusions were agreed in respect of the mobile phone licences. Allowing for updated terminology, these were:
 - 1. The spectrum constitutes a natural resource.
 - 2. The licence to use the spectrum constitutes an asset described as a permission to use a natural resource which is a sub-set of the general asset class of contracts, leases and licences.
 - 3. Typically licence payments are neither taxes nor purchases of the spectrum itself.

- Land, mineral deposits and the spectrum are similar types of assets and so are leases and licences based on the use of those assets.
- 5. There is no single, universal and clearcut criterion to distinguish between rent and asset sale; a range of criteria needs considering.
- 6. Most cases examined point to treating licence payments as the purchase of an asset, not rent.
- 7. The value of the licence and the value of the spectrum move symmetrically.
- 8. Further elaboration will be useful in future.
- 17.275 The considerations referred to under conclusion 5 were six in number and are reproduced below.
 - Costs and benefits assumed by licensee: the more of the risks and benefits associated with the right to use an asset are incurred by the licensee, the more likely the classification of a transaction as the sale of an asset as opposed to rent. Thus, pre-agreement on the value of payments (whether by lump sum or by instalments) effectively transfers all economic risks and benefits to the licensee and so point to the sale of an asset. If, on the other hand, the value of payment is made contingent on the results from using the licence, risks and benefits are only partially transferred to the licensee and the situation is more readily characterised as payment of rent. In the case of mobile phone licences, the total amount payable has often been preagreed. An additional indication of the degree to which commercial risks have been passed to the licensee is to examine the hypothetical case where a licensee goes bankrupt. If, in such a case, the licenser reimburses none of the up-front payment made by the licensee, this would constitute a strong case against a characterisation of the transaction as rent, as apparently the licensee has incurred all the commercial risks involved.
 - Up-front payment or instalment: as with other indicators, the mode of payment is in

- itself not conclusive for a characterisation as asset or rent payment. Generally, the means of paying for a licence is a financial issue and as such not a relevant factor in determining whether or not it is an asset. However, business practice shows that upfront payments of rent for long periods (15-25 years in the case of mobile phone licences) are highly unusual and this favours an interpretation as sale of an asset.
- 3. **Length of the licence**: licences granted for long periods suggest a treatment as the sale of an asset, for shorter periods a treatment as payments for rent. The time frame involved in mobile phone licensing (15-25 years) is considered rather unusual as a period for which to conclude a fixed payment of rent and therefore a further indication favouring an interpretation as sale of an asset.
- 4. Actual or de-facto transferability: the possibility to sell the licence is a strong indication of ownership and if transferability exists, this is considered a strong condition to characterise the licensing act as the sale of third-party property rights. In practice, mobile phone licences are often transferable either directly (by the business selling the licence to another business) or indirectly (through the business being acquired through a take-over).
- 5. Cancellation possibility: the stronger the restrictions on the issuer's capacity to cancel the licence at its discretion, the stronger the case for treatment as a sale of an asset. Conversely, when licences can easily be cancelled at the discretion of the issuer, ownership over benefits and risks has not been fully transferred to the licensee and the transaction qualifies more readily as rent.
- 6. Conception in the business world and international accounting standards: businesses, in accordance with international accounting standards, often treat a licence to use the spectrum as an asset. Again, in itself this does not lead to treatment as an asset in the national accounts, and there are other areas where

companies choose to present figures in their accounts in ways that are not consistent with the national accounts. But the treatment of the acquisition of mobile phone licences as capital investment in company accounts provides an added incentive to treat them in a similar way in the national accounts.

17.276Not all these considerations have to be satisfied to characterise the licence as a sale of an asset nor does a simple majority of them being satisfied do so. However, in order to qualify as a rental agreement, at least some of the following sorts of conditions should hold.

The contract is of *short-term duration*, or renegotiable at short-term intervals. Such contracts do not provide the lessee with a benefit when market prices for the leased asset go up in the way that a fixed, long-term contract would. Such benefits are holding gains that typically accrue to owners of assets.

The contract is *non-transferable*. Non-transferability is a strong but not a sufficient criterion for the treatment of licence payments as rent, because, although it precludes the lessee from cashing in on holding gains, it does not preclude the lessee from reaping comparable economic benefits (e.g., using the licence in their business).

The contract contains *detailed stipulations* on how the lessee should make use of the asset. Such stipulations are often seen in cases of rent of land, in which the owner wishes to retain a control over the usage of the land. In the case of licences, examples of such stipulations would be that the contract states what regions or types of customers should be served, or that it sets limits on the prices that the lessee may charge.

The contract includes conditions that give the lessor the *unilateral right to terminate* the lease without compensation, for instance for under-use of the underlying asset by the lessee.

The contract requires payments over the duration of the contract, rather than a large u-pfront payment. Although this condition is essentially financial in character and thus cannot be decisive on the type of the lease, it

may indicate a degree of control for the lessor to direct the use of the spectrum. The case for a treatment as rent is further supported if the payments are related to the revenue the lessee derives from the licence.

17.277 These two sets of considerations can be seen as a more specific parallel to the distinction of economic ownership from legal ownership used in distinguishing between an operating and financial lease as described above. conditions for treatment of the payment as the acquisition of an asset and for treatment as payment of rent are indicative rather than prescriptive. A decision on the appropriate treatment when some of the conditions are not met will necessitate consideration of how to record those conditions not met. For example, if on balance the decision is to treat the payment as rent but a large up-front payment was made, this should be treated as a pre-payment to be recorded on an accrual basis. However, if the recipient is not willing to consider a refund if the contract is suspended, accrual recording is difficult. This is one reason why up-front payments are often indicative of the sale of an asset rather than the payment of rent.

17.278The application of these principles to the main forms of natural resources are described below, beginning with radio spectra.

2. Radio spectra

- 17.279 Payment for a mobile phone licence constitutes the sale of an asset, not payment for rent, when the licensee acquires effective economic ownership rights over the use of the spectrum. To decide whether ownership is effectively transferred or not, the six criteria quoted above are to be considered.
- 17.280 When sale of an asset applies and when the life span of the licence and of the spectrum coincide, the payment for a licence is treated as the sale of the spectrum itself. The latter situation applies always when licences are granted indefinitely.
- 17.281 When sale of an asset applies, and when the life span of the licence is different from the life span of the spectrum, the payment for a licence is treated as the sale of a permit to use a natural resource by the legal owner (licensor) to the economic owner (licensee).

17.282When the licence agreement is treated as the sale of an asset in its own right, its value is established at the time of its sale. It declines with the expiration of the period of validity to fall to a value of zero at the point of the expiry of the licence. Symmetrically, the value of the spectrum falls when the licence acquires a value and is progressively re-established as the licence expires. This is consistent with a potential further sale of the right to use the spectrum for another period. This procedure also ensures a neutral effect on the net worth of the overall economy during the life of the licence.

3. Land

- 17.283Land may be sold outright when the legal ownership is transferred from one institutional unit to another. (Land may not be recorded as being sold to a non-resident unit. In such cases a notional resident unit is created that holds title to the land; the non-resident unit then owns the equity of the notional resident unit.)
- 17.284The type of asset most frequently subject to a resource lease is land. Tenant farmers usually pay regular rent to their landlord. A resource lease on land may be considered as a sale of the land if the lease satisfies most or all of the same criteria as those listed for payments for a mobile phone licence to be considered a sale of an asset. When the land is leased in other circumstances, the payments are recorded as rent under a resource lease agreement.
- 17.285In some jurisdictions, the land under buildings remains in the ownership of a landlord other than the owner of the buildings. If regular payments are made to the landlord, these are recorded as rent. However, it is sometimes the case that, even though the land technically belongs to another unit, when the building changes ownership, the purchase price includes an element representing the present value of future rent payments. In such a case, the land is recorded in the System as if the ownership is transferred to the owner of the building above the land. If, at the end of the land lease, a further payment is liable for extension of the lease for another long-term period, this should be recorded as capital formation and an acquisition of an asset in a manner similar to costs of ownership transfer on purchase and sale of an asset.

4. Timber

- 17.286If a unit is given permission to clear fell an area of natural forest, or to fell at its discretion without any restriction in perpetuity, the payments made to the owner constitute the sale of an asset. (The sale of forested land may be recorded as the sale of the timber and the land separately, depending on the intended use of each.)
- 17.287The option to have a lease permitting felling at the lessee's discretion but subject to the restoration of the land, in an acceptable forested state, at some time in the future is improbable. It is more common for timber felling to be allowed under strict limits with a fee payable per tree felled (stumpage). The limits are usually such that the harvest of timber is sustainable and so the payments are recorded as rent in the case of a natural forest.
- 17.288Forests may also be produced assets, in which case the extraction of timber is treated as the sale of a product.
- 17.289 Illegal logging across national borders is prevalent in some countries. In such cases the quantity of timber extracted should be recorded as uncompensated seizure of a natural resource or cultivated asset, as the case may be.

5. Fish

- 17.290 Natural fish stocks are an asset and the same considerations apply to them as to other natural resources. It is not realistic to consider that permission would be given to exhaust fish stocks but illegal fishing may either reduce the stock below the point of sustainability or exhaust them altogether. In these cases, uncompensated seizure of the stock should be recorded.
- 17.291 Fishing quota may be allocated in perpetuity or for extended periods to particular institutional units, for example, where fishing is an established way of life and there may be little alternative economic employment. In such circumstances the quota may be transferable and if so, there may be a well developed market in them. Fishing quota may therefore be considered as permits to use a natural resource that are transferable.

- 17.292 An alternative regime is to issue a permit for a strictly limited period of time, less than a year, to a nominated institutional unit, often a non-resident. This is a common practice in some islands in the South Pacific, for example. In these cases the revenue from the licences should be recorded as rent as under a resource lease.
- 17.293A licence for recreational fishing has long been considered, by convention, as payment of a tax. This treatment is not changed by the wider considerations for commercial fishing.

6. Water

- 17.294A body of water with an economic value can be sold in its entirety either as part of the land that surrounds it or as a separate entity.
- 17.295 As is the case for fish, it is unlikely that economic ownership would be ceded under a long lease with no preconditions on the quantity and state in which a similar amount of water should be returned to the owner. However, it is possible that surface water could be leased under a long lease for recreational purposes, say. The treatment of such leases should be as for land.
- 17.296Of increasing concern is the extraction of water from water bodies. Regular payments for the extraction of water (as opposed to the delivery of it) should be treated as a resource rent.

7. Mineral deposits

17.297 Mineral deposits differ from land, timber and fish in that although they also constitute a natural resource, there is no way of using them sustainably. All extraction necessarily reduces the amount of the resource available for the future. This consideration necessitates a

- slightly different set of recommendations for how transactions relating to their use should be recorded.
- 17.298 When a unit owning a mineral deposit cedes all rights over it to another unit, this constitutes the sale of the deposit. Like land, mineral deposits can only be owned by resident units; if necessary a notional resident unit must be established to preserve this convention.
- 17.299 When a unit extracts from a mineral deposit under an agreement where the payments are made each year dependent on the amount extracted, the payments (sometimes described as royalties) are recorded as rent.
- 17.300A full articulation of the balance sheets of the units owning and extracting the mineral deposit is difficult. The owner (in many but not all circumstances government) does not have a productive activity associated with the extraction and yet the wealth represented by the deposit declines as extraction takes place. In effect, the wealth is being liquidated with the rent payments covering both a return to the asset and compensation for the decline in Although the decline in wealth is caused by the extractor, even if the deposit were shown on the balance sheet of the extractor, the rundown in wealth would not be reflected in the extractor's production account because it is a non-produced asset and thus not subject to consumption of fixed capital. (The SEEA 2003 describes a form of satellite account where such a deduction from national income can be made for minerals as well as for other natural resources used unsustainably.) For these reasons, simple recording of payments each year from the extractor to the owner as rent and changes in the size and value of the deposit as other changes in the asset accounts is recommended.

P. Sharing assets

- 17.301There are two ways in which assets may be shared. The asset may be wholly owned by two or more units, each at different points in time. Alternatively, the risks of and benefits from the asset may be shared by two or more units at a single point in time. The two cases require different treatments.
- 17.302 Within the System, even though the asset may be owned by different units at different times, when a balance sheet is drawn up, the whole of the value of the asset is attributed to one unit. For an asset subject to an operating lease, there is no ambiguity. The legal owner is also the economic owner and is the unit that shows the

asset on its balance sheet. For an asset subject to a financial lease, the unit showing the asset on its balance sheet is the economic owner. This is consistent with the views that the value of the asset represents the stream of future benefits coming from the asset and the economic owner is the unit entitled to receive these benefits in return for accepting the risks associated with using the asset in production. For an asset subject to a resource lease, the value is shown on the balance sheet of the legal owner.

17.303 When a licence is issued to use a natural resource exclusively for a period of time less than the expected life of the asset, the natural resource remains on the balance sheet of the legal owner with a value set equal to the rents expected from the end of the lease forward. An exception is made for a land lease satisfying the six criteria established in the mobile phone case, where for practical reasons, the value of the land may be merged with the value of the buildings on it if it cannot be separately identified. In this case the value of the asset on the owner's balance sheet is the value of the land from the time that the lease expires forward. When licences to use other natural assets such as radio spectra, timber and fish satisfy the "mobile phone" criteria, a separate asset, described as a permit to use a natural resource, is established. These assets are part of the sub-class of contracts, leases and licences.

Licences to use natural resources that give rise only to rent do not affect the recording of the resource on the balance sheet of its owner.

17.304 Sharing the risks and rewards of an asset between different units at a point in time is unusual. The most common occurrence is that a single unit undertakes the activity in which the asset is used and that unit shares the returns among the owners in the form of distributed property income. However, occasionally it is possible such a single unit does not exist and it is not meaningful to try to create it statistically. This is most common when the participating units are resident in different economies, as may be the case with an airline, or in the case of some unincorporated joint ventures (UJVs). The terms under which UJVs are established are diverse but one form allows that all members share the assets equally. In such cases, the System records the assets shared between the owners in proportion to their ownership shares.

17.305 In some joint ventures, one party may contribute an asset as its share of the costs. If this happens, an injection of capital equal to the value of the asset should be recorded followed by the purchase of the asset in question with the ownership of the asset then shared by all parties to the arrangement.

Q. Permissions to undertake a specific activity

17.306 Permissions may be granted not only to use an asset but also to engage in a particular activity, regardless of the assets involved in the activity. The permissions are not dependent on a qualifying criteria (such as passing an examination to qualify for permission to drive a car) but are designed to limit the number of individual units entitled to engage in the activity. Such permissions may be issued by government or by private institutional units and different treatments apply to the two cases.

1. Permits issued by government

17.307When governments restrict the number of cars entitled to operate as taxis or limit the number of casinos permitted by issuing licences, they

are in effect creating monopoly profits for the approved operators and recovering some of the profits as the fee. In the System these fees are recorded as taxes, specifically as other taxes on production. This principle applies to all cases where government issues licences to limit the number of units operating in a particular field where the limit is fixed arbitrarily and is not dependent only on qualifying criteria.

17.308In principle, if the licence is valid for several years, the payment should be recorded on an accrual basis with an other account receivable/payable entry for the amount of the licence fee covering future years. However, if government does not recognise a liability to repay the licence-holder in the case of a

cancellation, the whole of the fee payable is recorded at the time it is paid.

17.309The incentive to acquire such a licence is that the licensee believes that he will thereby acquire the right to make monopoly profits at least equal to the amount he paid for the licence. This stream of future income is treated as an asset if the licensee can realise this by onselling the asset. The type of asset is described as a permit to undertake a specific activity. The value of the asset is determined by the future stream of monopoly profits. If the payment for the licence is being recorded by government on an accrual basis, the licensee has an asset in his balance sheet under accounts receivable/payable equal to the value of the future payments and so the value of the licence itself should cover simply the excess of the monopoly profits over the cost. If the licence is on-sold, the new owner assumes the right to receive a refund from the government if the licence is cancelled as well as the right to earn the monopoly profits. If the licence was recorded as a single tax payment, the value of the asset is estimated as the value of all the future monopoly profits without deduction. The asset first appears in the other changes in the volume of assets account and changes in value, both up and down, are recorded in the revaluation account.

17.310Governments are increasingly turning to the issuing of emission permits as a means of

controlling total emissions. These permits do not involve the use of a natural asset (there is no value placed on the atmosphere so it cannot be counted as an asset) and are therefore classified as taxes even though the permitted "activity" is one of creating an externality. It is inherent in the concept that the permits will be tradeable and that there will be an active market in them. The permits therefore constitute assets and should be valued at the market price.

2. Permits issued by other units

17.311It is less common for units other than government to be able to limit the participation in a given activity. One instance may be when it is either compulsory or desirable to belong to a professional association but in this case there is seldom a limit on numbers participating. Another example could be where the owner of property limits the numbers of units allowed to operate on his property for example a hotel with a policy of only allowing one firm of taxis to pick up guests. In these sorts of cases, the permits are treated as payments for services. In principle the payment should be recorded on an accrual basis throughout the period for which the permit is valid. There is no reason in principle why such permits could not be treated as assets if they were marketable though this may not be a common situation.

R. Contracts for future production

17.312 Although human capital is not recognised as an asset in the System, there are cases where a contract that entitles the holder to limit the ability of a named individual to work for others may be regarded as an asset. The most prolific and lucrative contracts may be for sports players where, for example, a football club can "sell" a player to another. In fact they are not selling the person, they are selling the exclusive right to have that person work for them. contracts exist for the rights to publish literary works or musical performances. All such contracts are treated as assets of the type entitlement to goods and services on an exclusive basis within the asset class of contracts, leases and licences.

17.313It is possible to imagine that similar contracts may exist for the production of goods in future. An examination of the practice of purchasing the options of future aircraft production revealed, however, that in this case there is no transferable asset and a change of mind on the part of the potential purchaser or failure to deliver on the part of the supplier is settled by a change in the arrangements between the two parties and does not lead to the sale of the option to a third party. If an instance arises where the option to purchase goods is treated in the same way as a contract for a named individual's performance, the same classification would apply.

S. Leases as assets

- 17.314As stated at the beginning of this section, contracts underlie many transactions recorded in the System and it is important to understand what the implications are for the time of recording and classification of transactions arising from a contract. It has been noted that permits or licences to use natural resources may constitute an asset as may permits to undertake specific activities and contracts for future production. There is one other condition that may lead to a contract being considered as an asset, which is another circumstance when the contract is transferable to a third party (that is a unit other than the two specified in the original contract).
- 17.315 Suppose a lease on an apartment agreed some time ago specifies the rental at 100 per month but if the same apartment were to be leased currently it would fetch 120 per month. From the lessor's point of view, the apartment is "encumbered" by the existing lease, that is, it carries a penalty (in this case of 20 per month) because of the existence of the lease. The encumbered value of the apartment is based on the present value of future rental payments taking the existence of the lease into account, that is, the future income stream is 100 for as long as the lease lasts and 120 thereafter (ignoring any allowance for inflation). unencumbered value of the apartment is a present value based on an income stream of 120 per month from the current period forward. The value to be entered in the landlord's balance sheet is the encumbered value. If he wishes to sell the apartment and the existing tenant had the right to remain at the agreed rental, the encumbered value is all the landlord (lessor) could hope to realise. If he wished to realise the unencumbered value he would have to pay the tenant the difference between the unencumbered value and the encumbered value to be free of the lease. This amount, the encumbrance, can in some circumstances be treated as an asset of the tenant. The circumstances are that it is both legally possible and is practicable for the tenant to sublet the apartment to a third party. Because of the difficulty of identifying when such assets may exist, it is recommended that in practice

- these assets be recorded only when there is evidence that they have been realised.
- 17.316It is possible that the encumbered value of the apartment may be higher than the unencumbered value if rentals have fallen since the lease was agreed. In this case it is the landlord that has a potential asset. The value of the apartment in his balance sheet is still the encumbered value. If the tenant wishes to cancel the lease, he may have to pay the landlord the difference between the encumbered value and the unencumbered value. Once the lease expires or is cancelled, the value of the apartment returns to its unencumbered value.
- 17.317 Assets reflecting such third-party property rights are always transitory. They exist only for the length of the lease and where there is a difference between the encumbered and unencumbered values. As each year passes, they reduce in value because the period during which the difference exists is reduced but may increase if the new rental price increases.
- 17.318The market price of the rental of an apartment is the price actually paid by the existing tenant. If, in this example, the original tenant remains in situ and pays 100 per month, this is the market price despite the fact that a new lease would fetch a rental of 120. Only if the original tenant sublets the apartment for 120 would the market price be recorded as 120. Of this, 100 will be paid to the landlord and 20 to the original tenant.
- 17.319The example above shows when a marketable operating lease may acquire a value as an asset. Permits to use natural resources and contracts for future production may also give rise to these sorts of third-party property rights assets. So may permits to undertake specific activities even though the original payment was treated as a tax if payable to government. Financial leases do not give rise to these sorts of assets. If the value of the asset being leased increases by more than the payments due under the financial lease, the lessee always has the option of selling the asset, repaying the loan and keeping the difference.

T. Other considerations

1. Time share arrangements

- 17.320One way of sharing an asset offering accommodation is by means of a "time share" arrangement. The same expression, though, may be used for a number of different arrangements.
- 17.321 One arrangement is similar to purchasing a house except that "ownership" is restricted to a particular period each year but in perpetuity. Exactly the same physical space is available to the owner each year. Another arrangement guarantees accommodation at a given time each year but not necessarily in the same physical space. Other arrangements consist of buying "points" in a scheme that the owner can use to purchase accommodation at different locations and times subject to availability.
- 17.322 All time-share arrangements have a unit that is responsible for upkeep, maintenance, insurance and so on but there are variations in whether this unit is the ultimate owner of the complex and the subscribers are lessees or whether the unit acts as agent for the group of owners/subscribers. Similarly there are variations in whether the owner/subscriber may sell or bequeath his ownership to another unit permanently and whether they can sublet occasionally.
- 17.323The issue of whether participation in the timeshare scheme gives rise to an asset will depend on the answers to these sorts of questions. If the owner has a nominated space, available in perpetuity, is eligible to act as part of the management committee for the scheme, can sell or bequeath the allocation at will, then the holding is most likely to be an asset of the same type as a house. If the owner has a fixed

- agreement to have some form of accommodation available at a given period for a fixed length of time, it is likely that this represents a pre-paid lease but one that could be sub-let occasionally or sold for the rest of the period of the lease as a transferable operating lease. A participant in a point based schemes may have only an account receivable by way of an asset.
- 17.324Where time-share arrangements are significant, the conditions pertaining to them should be examined in the light of the general principles described in this section to determine how to record the transactions involved and classify the assets.

2. Lost deposits

17.325 Under any form of contract, it is possible that one party makes a payment and the other does not deliver the goods, services or assets promised in the contract. In many cases this gives rise to an account payable/receivable that the first party may reclaim from the second. In some circumstances this may not be possible. For example, cheap airline tickets are often offered on a non-refundable basis. The fact that pre-payments are non-refundable is part of the business plan of the company concerned. Their output should be measured as the value of sales without reduction for the payments by clients who did not avail themselves of the services to which they were entitled. Volume measures of output will depend on the services actually delivered and the impact of the non-refundable deposits will show up as a price effect. It will also be reflected in the consumption expenditure figures of those paying for services they did not in the end take delivery of.